Published in *Journal of modelling in management*, to be published, which should be cited to refer to this work. DOI : 10.1108/JM2-10-2023-0238

# The Intention of Adopting Blockchain Technology in Agri-Food Supply Chains: Evidence from an Indian economy

# Aditi Saha

Department of Operations and Supply Chain Management, Indian Institute of Manageemnt (IIM), Mumbai, Vihar Lake, NITIE, Powai, Mumbai, Maharashtra, India. Pin- 400087 Email: aditi.saha.2020@nitie.ac.in

# Rakesh D. Raut

Department of Operations and Supply Chain Management, Indian Institute of Manageemnt (IIM), Mumbai, Vihar Lake, NITIE, Powai, Mumbai, Maharashtra, India. Pin- 400087

Email: <u>rraut@nitie.ac.in</u>

# **Mukesh Kumar**

Department of Mechanical Engineering, National Institute of Technology (NIT) Patna, India 800005 Email: <u>mukesh.me18@nitp.ac.in</u>

## **Sanjoy Kumar Paul**

UTS Business School, University of Technology Sydney, Sydney, Australia Email: <u>sanjoy.paul@uts.edu.au</u>

# Naoufel Cheikhrouhou

Department of Geneva School of Business Administration HES-SO Rue de la Tambourine 17, 1227 Carouge, Switzerland Email: <u>naoufel.cheikhrouhou@hesge.ch</u>

Corresponding email: <a href="mailto:rraut@nitie.ac.in">rraut@nitie.ac.in</a> (R.D. Raut)

#### Abstract

**Purpose** -This research explores the underlying intention behind using blockchain technology (BLCT) in the agri-food supply chain (AFSC). This is achieved by employing a conceptual framework based on Technology Acceptance Models (TAM) that considers various factors influencing user behavior towards implementing this technology in their practices.

**Methodology-** The conceptual framework developed is empirically validated using Structural Equation Modeling (SEM). A total of 258 respondents from Agri -food domain in India were involved in this survey, and their responses were analyzed through SEM to validate our conceptual framework.

**Findings-** The findings state that food safety and security, traceability, *transparency*, and *cost* highly influence the intention to use BLCT. Decision makers of the AFSCs are more inclined to embrace BLCT if they perceive the usefulness of the technology as valuable and believe it will enhance their productivity.

**Implications-** The study contributes to the existing literature by providing thorough examination of the variables that influence the intention to adopt BLCT within the AFSC. The insights aim to benefit industry decision-makers, supply chain practitioners, and policymakers in their decision-making processes regarding BLCT adoption in the AFSC.

**Originality/Value-** The current study investigates how decision-makers' perceptions of BLCT influence their intention to use it in AFSCs, as well as the impact of the different underlying factors deemed valuable in the adoption process of this technology.

**Keywords:** Blockchain (BLCT); Food security and safety; Agri-food supply chain (AFSC); Technology Acceptance Models (TAM); structural equation modelling (SEM)

## 1. Introduction

Food and agriculture are crucial for the Indian economy, but traditional technologies remain prevalent. With an estimated population of 9.6 billion by 2050, the food supply must increase by 70% (FAO 2009; Gardas et al.,2017). Blockchain technology (BLCT) is a potential solution to maintain traceability and transparency in the agri-food supply chain. Modern companies require transparency in food product history and secure information throughout the supply chain (Dabbene et al. 2014). Globalization and increased competition have strained the agri-food supply chain, leading to increased competition and a more complex supply chain. BLCT can help provide proof and authenticity of sources. BLCT can initiate many changes in the industry (Agrawal et al., 2023). It can change the business models of existing firms, change the level and nature of demand, affect the competitive positions of various players in the industry, and have a profound impact on the market (Lanzolla et al., 2012; Saurabh et al., 2023). Technology has proven effective as information moves through the SC (Guggenberger et al.,2020; Ali et al., 2021). As a new decentralized and distributed technology, BLCT can transform a wide range of industries by enabling transactions to be recorded and verified in a secure, transparent, and immutable manner. It can address major supply chain management (SCM) issues like visibility, traceability, and transparency (Ahmed et al., 2022). The concept of Blockchain for Good (B4G) gives rise to novel applications that can drive the United Nations Sustainable Development Goals (Trollman et al., 2022).

Integrating BLCT in AFSC can provide innovative solutions for product traceability and food safety (Feng et al., 2020; Kumar et al., 2023a). As a result, many of the current challenges of food safety, security, and transparency faced by SCs can be resolved (Gardas et al., 2019). The immutable and decentralized nature of the BLCT helps to eliminate information asymmetry in SCs by storing data in the form of blocks, enabling the storage and trading of tangible and intangible assets with increased efficiency, reducing risks, and lowering costs (Li. et al., 2021). By providing direct linkages between farmers, retailers, customers, and producers, BLCT helps to restructure AFSCs, improving food quality and transparency and minimizing food fraud and wastage (Li et al., 2021; Collart and Canales, 2022). According to Wang et al. (2021), a BLCT tracing system increases SC transparency and process management and reduces intermediary costs, improving SC service and confidence. With the importance of Blockchain technology in advancing AFSC to greater transparency and traceability, it still faces issues in adoption; thus, the research question still needs to be addressed: *What factors are perceived as helpful in adopting the BLCT in AFSC*?

Successful adoption of BLCT in AFSC will confer a competitive edge and unparallel opportunities to the sector. A significant number of practitioners and scholars point out the BLCT's ability to transform SCs (Schmidt and Wagner, 2019; Wang et al., 2019b; Chang et al., 2020; Wamba et al., 2020) and (Zhang et al., 2020; Ali et al., 2021; Mangla et al., 2021; Vikaliana et al., 2021) studied BLCT in many agri-food industries to understand the implementation benefits and challenges. However, the scarcity of information and an inability to apply BLCT to widely spread applications have hindered such applications, specifically in developed economies. Also, gaining acceptance from key stakeholders and decision-makers remains a pivotal factor in successfully implementing such innovations. Therefore, it is critical to establish the intention to use BLCT in AFSC to understand the

factors influencing its adoption and develop strategies for promoting its successful implementation. And thus, this study intends to address the second research question: *How do the factors identified influence the intention to adopt BLCT in AFSC?* 

Integrating BLCT into the AFSC requires user acceptance and understanding of the factors they perceive as essential for adoption. To ensure the benefits of BLCT, AFSC decision-makers must ensure it benefits their organization and secures consumer trust. To achieve this, a comprehensive assessment of user perceptions and attitudes towards the new technology is necessary (Saha et al., 2023; 2023a). Over time, numerous studies have explored this topic with varying degrees of detail, including seminal works authored by (Venkatesh et al., 2003; 2012; Queiroz and Wamba, 2019a). The Technology Acceptance Model (TAM) is a theoretical framework used to evaluate users' willingness to accept and utilize new technology. The original TAM model, proposed by Davis in 1989, has evolved into different versions, each with unique features. It focuses on perceived usefulness and ease of use, making it a straightforward and practical model for researchers and practitioners studying and improving user acceptance of technology.

User acceptance of the BLCT and its influencing factors, play an important role in the technology's adoption. By considering different technology adoption models and independent variables that influence user acceptance of BLCT, it is possible to predict the level of user acceptance of BLCT in the AFSC domain. Furthermore, it is crucial to understand the decision makers' intention in adopting BLCT, considering this technology's disruptive potential (Wamba and Queiroz 2020) and its unprecedented impact on SCs. Though recent research (Yadav et al., 2020; Liu et al., 2021; Trollman et al., 2022; Kumar et al., 2023a) investigated the potential and factors of BLCT in AFSC, the factors contributing to its widespread acceptance and intention to acknowledge the technology have been overlooked. While few authors (Kamble et al., 2019; Nayal et al., 2021; Paul et al., 2021) emphasized its impact on SC, they did not identify the key factors influencing decision -makers' intentions for BLCT adoption, resulting in gaps in understanding the SC dynamics and stakeholders' perspective. Thus, this study investigates the factors influencing decision-makers intentions to use BLCT in AFSC. The study conducts empirical research using SEM and proposes a conceptual framework based on TAM to study the intention to use BLCT and shed light on the following research objectives (ROs):

*RO1: To explore the factors helpful in adopting the BLCT in AFSC. RO2: To examine the user intention to adopt BLCT in AFSC.*  Since blockchain is still in the inception stage and adoption of the technology remains a challenge, this research delves deeper into identifying and understanding the factors that influence the adoption and intention to use BLCT in AFSC. To achieve this, the study is based on TAM, which prioritizes perceived usefulness (PU) and perceived ease of use (PEU) as predictors of intentions to use (ITU). TAM is used to build the study's conceptual framework, which identifies critical factors impacting ITU while also studying their effects on the TAM construct to gain better insights toward successful implementation. These findings will enable the decision-makers to develop more robust strategies to integrate BLCT across the AFSC effectively. By understanding the factors that drive technology adoption, tailored implementation plans can be fostered for the Indian AFSC.

The research paper is laid out in the following pattern. Section 2 discusses the overview of the relevant literature. Section 3 presents the conceptual framework's development and the formulation of the hypotheses. Section 4 addresses the research design. In Section 5, the data analysis and findings are presented. Section 6 includes a discussion and theoretical and practical contribution of the study. Finally, section 7 concludes the paper with a discussion on future directions.

#### 2. Literature Review

#### 2.1. Blockchain technology and AFSC

Integrating BLCT in AFSC offers a comprehensive solution to trace, track and monitor the entire length of the supply chain (Kayikci et al.,2022; Srivastava et al., 2022), ultimately leading to greater transparency and trust (Liu et al., 2022; Sharma, 2023). While traditional SC have long struggled with food safety concerns, BLCT can address these issues by increasing visibility throughout the chain (Kshetri 2019). Research studies conducted on BLCT and SC integration revealed that adopting this technology improves transparency, authenticity, and real-time transactions (Mukherjee et al., 2021; Feng et al.,2020; Xu et al.,2020). Furthermore, (Sunny et al., 2020) suggest that transparency and traceability are critical factors that impact a logistics organization's overall performance – making it essential for businesses to prioritize integrating BLCT into their SCs. Providing complete visibility upstream and downstream of AFSC-BLCT helps establish accountability across all supply chain stages while enhancing its trustworthiness (Rogerson and Parry, 2020). More and more value chains are incorporating BLCT to make them flexible and strengthen customer ties (Murki et al., 2018). This is a major movement in the rapidly expanding food industry, where BLCT is seen as the best option for tracking and changing. The potential uses of BLCT in the agri-food value chain have been the subject of several studies (Zhao et al., 2019).

One such study used RFID and blockchain to create a system to track the agri-food supply chain in China (Tian. 2016).

Further applications of blockchain technology include an IoT software connector, a distributed ledger manufacturing supply chain that is blockchain-ready, and a supply chain traceability solution that is both transparent and decentralized. By facilitating the transfer of more precise and trustworthy data between producers and consumers, BLCT has shown to be more effective than conventional approaches in food supply chains (Zhao et al., 2019). Because of this, the food sector is now more efficient and cost effective. Ultimately, the food sector is realizing the importance of BLCT, which provides a safer and more effective method to monitor and record food supply chains. Businesses may improve their operations and cut expenses by using BLCT in several industries, including supply chains, logistics, and agriculture.

As the research shows, integrating BLCT in AFSC is an innovative move that promises many advantages. With its ability to improve transparency and trust throughout the SC, BLCT has been proven beneficial for AFSC. However, it is worth noting that despite this promising development, the intention to use BLCT in AFSC is still relatively new and requires further exploration before its potential benefits can be fully realized. For decision-makers to effectively integrate BLCT into their AFSC, they must first believe in its perceived usefulness and ease of use of the technology. This requires a deeper understanding of how BLCT works, and which factors benefit and fit within their specific industry context.

#### 2.2. Research Gaps

BLCT adoption and intention to use in SC has been studied by various authors such as Wamba & Queiroz (2020) in the Brazilian SC, Queiroz et al. (2021) in logistics and SC, Karamchandani et al. (2022) for the service SC, Sternberg et al. (2021) in inter-organizational SC. While several models on BLCT adoption have emerged in recent years within the agriculture and food sectors context, studies such as (Saurabh and Dey, 2021; Ronaghi, 2021; Paul et al., 2021; Liu et al., 2021; Yadav et al., 2020; Susanty et al., 2021), did not fully explore critical issues that drive its widespread acceptance. Queiroz et al. (2021) conducted a comprehensive study on adopting BLCT, predicting the likelihood of implementing BLCT, while Queiroz and Wamba (2019) emphasized that performance expectancy is one of the enabling conditions of BLCT adoption. Interestingly enough, both US-based firms and Indian organization were found to be influenced by this factor. Furthermore, Wong et al. (2020a) focused on additional considerations related to adopting BLCT - specifically costs incurred during the implementation phase, regulatory support from authorities, top management

involvement & encouragement towards its utilization within organization frameworks. The results indicated that cost implications had widespread influence over intentions regarding adoption rates among respondents surveyed. Several scholarly studies have delved into the multifaceted ways in which BLCT can impact and influence the performance of SCs, as observed by Nayal et al. (2021), Masudin et al. (2021), and Paul et al. (2021). Vasan et al. (2023) has explored the farmers' belief in adopting technology, but the entire supply chain has not been discussed. These investigations shed light on a complex interplay between BLCT and SC but fail to uncover the potential factors that are perceived useful by decision makers and lead to the intention to use the technology. Also, the intention to use BLCT in Indian AFSC has not been explored much through TAM.

Although few studies have been conducted on the potential for blockchain adoption in supply chain management, a more comprehensive analysis of users' intentions to use BLCT is necessary. Understanding the factors that motivate users to embrace this technology will be crucial in driving its widespread adoption throughout AFSC. To truly integrate blockchain as a solution, it is essential to examine BLCT's perceived usefulness and user motivation. To shed light on these key aspects of blockchain implementation, empirical research has focused on ITU BLCT in AFSC and variables deemed useful by prior literature. Notably, cost (COS), food safety and security (FSS), transparency and traceability (TT) have emerged as salient concerns related to adopting this technology within supply chains. A TAM framework has been adopted to comprehend how independent variables affect both PU and PEU of the technology, ultimately leading towards higher rates of ITU-BLCT in AFSC.

#### 3. Hypotheses Development and Conceptual Framework

## 3.1 Technology Acceptance Models (TAM)

Technological advancements continue to be significant in business, and acceptance of technology is required to implement them successfully. In light of this reality, companies must deeply understand the factors contributing to successful implementation. To elucidate these complexities, this study relies on TAM, originally proposed by Davis (1985), as an instrumental tool for assessing technological adoption. Extensive research has been conducted exploring different aspects of TAM over the years, including but not limited to the works by (Davis, 1985, 1989; Venkatesh and Davis, 2000; Venkatesh et al., 2003; Venkatesh and Bala, 2008; Venkatesh, Thong and Xu, 2012). By examining individual or organizational reasons behind technology uptake through applying the TAM framework, we can gain insight into how best to facilitate its utilization within businesses.

TAM has long been a cornerstone in the field of technology acceptance research. Although numerous variations, such as TAM2 and extended TAM models, have emerged, the original model

still holds significant weight due to its extensive validation. Venkatesh and Davis (2000) argue that while newer versions introduce new variables like social influence and cognitive instrumental processes, their complexity may limit practical application. This debate was put to the test by Wu and Wang's (2005) study comparing the usefulness of both original TAM versus extended version- results showed that although expanded versions explained more variance in data, the simplicity provided by the original model made it an effective tool for predicting technology acceptance patterns with reasonable accuracy. Ultimately, this affirms support for utilizing original TAM as an undeniably successful methodology for seeking insights into understanding how users accept new technologies effectively under varying contexts.

TAM explores the link between PU, PEU, and ITU characteristics. Using TAM, Davis et al. (1989) proposed that PU, and the attitude toward use, influence technology adoption. PU assesses an individual's attitude towards technology. This attitude usually reflects how much benefit they expect from using the technology and how much effort they think they will have to expend to use it. Furthermore, it is anticipated that PEU has a direct effect on PU. PEU represents how easy it is for an individual to use the product/service. A product with a high PEU score will be perceived as easy to use and therefore more likely to be adopted by users. ITU is the willingness of an individual to use a technology once the individual is aware of it. It is determined by measuring the level of motivation to use the technology in question as well as a person's attitude toward using such technology.

#### 3.2 Conceptual framework

The conceptual framework has been adapted from Venkatesh et al. (2003). The framework based on the TAM concept studies the perceived usefulness (PU) of the BLCT in the AFSC. The framework consists of three variables: COS, FSS, and TT, identified from the literature review as influential on the adoption and ITU the BLCT technology. Figure 1 shows the proposed conceptual framework. While the model is comprehensive, it may not cover all factors influencing the adoption of BLCT. However, to ensure parsimonious measurement instrumentation and considering the survey length, the study was limited to only three variables FSS, COS, and TT.



<sup>&</sup>quot;\*COS- Cost; FSS- Food safety and security; TT- Transparency and traceability; PEU- Perceived ease of use; PU- Perceived usefulness; ITU-Intention to use"

## 3.1 Cost

The cost of adopting a new technology goes beyond just the financial aspect. It also includes emotional effort, time, and energy investment to learn and integrate it into existing systems. However, this initial expense can eventually save money by streamlining operations and reducing transactional costs in the long run. Decision-makers are more likely to invest in technologies that will significantly reduce operational expenses over time. One such innovative solution proposed by Li et al. (2021) is using BLCT for automation, which eliminates manual involvement, intermediaries, and paperwork, thus minimizing costs associated with supply chain management. Incorporating smart contracts further helps optimize procurement processes as suppliers receive prompt payment upon delivery confirmation while reducing human interaction between parties. Vu et al., (2021) demonstrate that such automated procedures streamline workflows and cut down on transaction expenditures like distribution expenses or data encryption fees, thereby boosting overall efficiency levels. A recent study conducted by Du et al. (2020) found that adopting blockchain technology can lead to improved efficiency and cost reduction. Building on this, Ko et al. (2018) discovered that the costs associated with implementing new technologies directly impact users' perception of usefulness, ultimately influencing their decision to adopt or reject these innovations. However, the perceived cost might not directly influence users' beliefs about the usefulness of the technology or their intention to use it. These aspects are more related to the perceived benefits and value that the technology provides. The influence of cost is mediated through the perceived ease of use of the technology rather than directly impacting perceived usefulness or intention (Luarn et al., 2005; Amoako-Gyampah, 2007; Yen et al., 2010). So, we propose the hypothesis:

H1. Cost positively influences the PEU of BLCT.

## 3.2 Food Safety and Security

Food safety and quality assurance are more challenging with the increase in the global flow of goods (Fischer, 2013). Food safety and security methods include handling, processing, preparing, and storing meals in a way that reduces the likelihood of people becoming ill due to contaminated foods. Thus, FSS refers to delivering a product that has not been contaminated, damaged, or tampered (Marucheck et al., 2011). BLCT helps make the FSC more visible, traceable, and accountable, ensuring safety and security (Gupta and Shankar, 2023). BLCT minimizes food fraud by ensuring

that records are transparent and traceable (Danese et al., 2021). As financial records for all operations are documented and stored in the chain, it can also speed up auditing and conflict resolution (Chang et al., 2020). In addition, the BLCT allows parties to update product information in near real-time, minimizing the possibility of cross-contamination and improving responsiveness. It also assists in providing food safety status in real-time to all SC decision-makers, thereby providing a secure, distributed, transparent, and collaborative information system. In their research, Roy et al. (2020) state that food safety and security are essential in earning SC decision-maker trust. Trust leads to the PEU of technology, thereby increasing the PU and ITU the technology leading to the following hypotheses:

H2. Food safety and security positively influence the PEU of BLCT.
H3. Food safety and security positively influence the PU of BLCT.
H4. Food safety and security positively influence the ITU BLCT for AFSC.

## 3.3 Traceability and Transparency

SC traceability and transparency has become increasingly important to sustainable management (Mollenkopf et al.,2022) A traceable and transparent SC demonstrates that an organization is honest and upfront about how it conducts business (Kafetzopoulos et al., 2023) A blockchain-based traceable SC increases visibility, improves quality controls, and reduces risk (Chavalala et al., 2022). Transparency builds trust among suppliers, companies, and customers. Astill et al. (2019) state that traceable and transparent food production systems must facilitate data exchanges between stakeholders as this will lead to expediting processes and a reduction of transaction times. Kamath (2018) discusses how BLCT helps trace and identify the SC's origin and path and build SC decision-maker trust. Being an immutable technology, BLCT reduces human intervention and is thereby helpful in food recall and food wastage (Duan et al., 2020). The traceability and transparency benefit of BLCT leads to the PU and ITU of the technology. Hence, we suggest the following hypotheses:

H5. Traceability and transparency positively influence the PEU of BLCT.

H6. Traceability and transparency positively influence the PU of BLCT.

H7. Traceability and transparency positively influence the ITU BLCT for AFSC.

# 3.4 Intention to Use

Perceived adoption intention refers to a user's active intention to adopt a specific technology (Queiroz & Wamba, 2019). Among the main strands of the technology management literature, an important focus is on adoption intention, which explains how individuals or decision-makers respond to a given technology that may lead to its actual usage. Research has shown that adoption intention

is one of the most significant predictors of technology usage (Venkatesh et al., 2012) and competitiveness (Kshetri, 2018). In their research, Davis et al. (1989) define perceived usefulness as the degree to which a user believes employing a given framework would improve productivity and efficiency. The literature shows that perceived usefulness significantly influences users' attitudes and intentions. Currently, BLCT is being used in a wide range of sectors. The users are more likely to be positive towards BLCT if they perceive it as useful and can increase efficiency. "Perceived ease of use" refers to how easily someone believes a particular technology or system is used (Davis, 1989). The literature shows that PEU positively affects PU (Venkatesh and Davis, 1996). Increasing user perceptions of BLCT as useful will make users more positive (Liu et al., 2021). Maintaining competitiveness, easy tracking and tracing of products, monitoring fair trade, cooperating with multiple participants, and building SC decision-maker trust may influence the use of BLCT for achieving benefits for firms and lead to the ITU of the technology. Hence, the followings hypotheses have been formed:

H8. PEU positively influences the PU of BLCT.H9. PU positively influences the ITU BLCT for AFSC.

# 4 Research Design

# 4.1 Questionnaire Design

The study uses in-depth interviews and questionnaires to examine the impact of different factors on the intention to use BLCT. The survey method is used to elicit information about respondents' backgrounds and attitudes as well as their intentions for using BLCT. A five-point Likert scale ("strongly agree to strongly disagree") was selected to measure the items and variables. The survey questionnaire was designed based on the previous literature and were validated by experts (industry and academician). Six variables and 24 items were used in the survey. The constructs utilized in the questionnaire are listed in their sources in Table 1. Six industry and 4 academic experts working in the agri-domain were asked to evaluate the questionnaire and check each construct for its reliability and validity. They reviewed the items for ambiguity, consistency, and relevance to the survey (Kumar et al.,2022). In addition to ensuring the consistency and reliability of the questionnaires for large-sample surveys, the validity and reliability of each scale are also checked.

Variable		Item	Descrip	Refe	erences	
Cost			"The elements ind	icate the entire	"Tapscot	tt et al.,
	COS1	Adoption cost	expenses linked	l with the	(2018);	Creydt and
		_	implementation	of BLCT. It	Fischer,	(2019);

 Table 1: Variables and Items Table (Authors own creation)

	COS2	Technology cost	helps in reducing the overall AFSC expenses by minimizing the middle entities. It also helps	Ivanov et al., (2019); Yadav and Singh (2020): Wong
	COS3	Maintenance cost	in reducing the operating cost, manpower cost, and	et al., (2020a); Sternberg et al.,
	COS4	Transaction cost	maintenance costs via investing less in consumable items, compact data storage, sharing database, faster payments, less human interaction, rapid decision-making, strong management, and an efficient SC."	(2021); Vu et al., (2021)."
	FSS1	Improve food quality	"Food Safety and Security suggest that BLCT is an	
	FSS2	Reduce cross	effective technology to avoid food fraud, reduce cross	
Food Safety	FSS3	Maintain records	contamination, and increase trust in the AFSC. BLCT helps	"Galvez et al., (2018); Tan et al.,
Food Safety and Security	FSS4	Ensure security	to retrieve the required information, confine contaminated products from suppliers, and minimize the products recall range. BLCT also provides data security from cyber threat and food fraud. The data are stored in immutable blocks and cannot be tampered."	(2018); Astill et al., (2019); Creydt and Fischer, (2019); Duan et al., (2020); Lin et al., (2020)."
	TT1	Reduce food wastage	"BLCT helps to manage inventory, transportation, and	
	TT2	Reduce food fraud	tracking of products. Firms can track the information on the	
Traceability	TT3	Easier food recall	storage life of food products better thereby improving profits.	
and Transparency	TT4	Improved tracking	BLCT can also assist in promoting economic	"Kamath, (2018); Astill et al., (2019);
	TT5	Increase in SC decision- maker trust	sustainability of the AFSC by reducing food waste and food recalls. It shortens the time to process the food and track the inaccessible data which can be utilized to improve SC procedures and build SC decision-maker trust."	Jen Yin Yeh et al., (2019)."
Perceived	PEU1	ease of use	"BLCT contributes to address the complex industrial systems	"Venkatesh et al.,
ease of use	PEU2	ease of understanding	(CISs) challenges. BLCT reduces the intricacy of cash data	Christopher Lee et

	PEU3	Compatible to use	storage and recovery system of AFSC, inventory, marketing, finance, and other departments. Operational usefulness has a positively influence on the BLCT's ease of use."	al., (2019); Queiroz et al., (2019)."	
	PU1	Improve performance	"BLCT enhances the AFSC's	"Variath (2018).	
Perceived	PU2	Simplify process	performance and simplifies the complex AFSC network by	Christopher Lee et	
usefulness	PU3	Improve safety and security	tracking and tracing the entire SC. Thereby improving safety and security. It also maintains		
	PU4	Increase traceability	and stores records in an Gao et al., immutable form which is easy to Var et al. (20)		
	PU5	Maintain records	access but difficult to tamper."	v u ct al., (2021).	
	ITU1	Innovative technology	"BLCT is an innovative distributed technology that helps	"Kamath, (2018); Kamilaris et al.,	
Intention to use	ITU2	SC decision- maker satisfaction	build trust, reliability, and satisfaction among the SC decision-maker as it can provide	(2019); Kshetri, (2019); Lin et al., (2020);	
	ITU3	Better transaction facility	farm to fork data. BLCT helps to build a better and smooth- running transaction facility."	Schinckus, (2020); Wang et al., (2020)."s	

# 4.2 Sampling and data collection

Data are collected from the questionnaires provided to Indian agri-food practitioners working in the domain for at least 2.5 years and having a basic knowledge of BLCT. Data are collected between June and August 2022 from multiple agro-based Indian companies to generalize the findings to a broader range of companies (Nayal et al., 2021). The purposive sampling technique is used as the sampling method to identify the interviewee. Respondents are also addressed individually whenever possible to enhance the survey response rate. Five hundred professionals from the agribased industry, including vegetables and fresh fruits, dairy, and beverage industry, knowing the technology implementation benefit, are invited to participate in the study. Of 550 professionals, 258 provided useful answers (a response rate of 46.9%). The average participant's ages are between 36 and 55 years old. It has been observed that the male respondents outweigh the female respondents, with the majority of participants aged 25 to 75 years. The participants' demographic characteristics are categorized in "Table 2".

Table 2	"Demographic	profile of the res	pondents" (N	-258) (	(Authors own c	reation)
---------	--------------	--------------------	--------------	---------	----------------	----------

Items		N	Percentage
Age	25-35	96	37.2

	36-55	104	40.3
	56-75	58	22.5
	Total	258	
Gender	Male	140	54.26
	Female	118	45.73
	Total	258	
Educational	UG	98	38
Qualification	PG	107	41.5
	PhD	53	20.5
	Total	258	
Years of Experience	0-5	87	33.7
_	5-10	58	22.5
	10-15	62	24
	15-20	51	19.8
	Total	258	

## 4.3 Non response and common method bias (CMB)

Non-response bias was mitigated by employing exploratory and subjective approaches (Armstrong and Overton, 1977). To evaluate the difference in means for every scale, the t-test is conducted for the early and late responses of the participants for a non-response bias test (Karamchandani et al., 2020). No significant difference in the scale items for each construct was found for this study. To prevent the common method bias (CMB), the questionnaire was conscientiously designed and tested with expert practitioners. Anonymity was ensured during data collection, and respondents were assured that there were no right and wrong answers, emphasizing the significance of truthful answers (Podsakoff et al., 2003). It is important to check and control CMB when the data from all the constructs in a model are collected from a single respondent (Podsakoff and Organ, 1986). To minimize the effect of common method variance, the sequences of the survey questions for every respondent were randomly assigned, and the respondents were asked to give honest feedback while they were also assured that their responses would remain anonymous (Podsakoff et al., 2003). In addition, Harman's single-factor analysis was performed to evaluate and mitigate the potential presence of biases. The results showed a single factor of less than 40% of the variance, suggesting no CMB. Even though common method bias affects the linear relationships, it seems to have fewer effects on the analysis of interactions between variables.

## 5. Data Analysis and findings

SEM is the statistical method used for hypotheses testing. With SEM, one can validate the suitability of TAM in analyzing BLCT's ITU. SEM helps specify, estimate, and evaluate models of linear relationships among variables in relation to a relatively small number of unobserved variables

(Shah et al.,2006). The data analysis was carried out using AMOS software, which has all the tools necessary for creating and examining SEM path diagrams (Nayal et al., 2021) and analyzing SPSS files (Mangla et al., 2020). IBM-SPSS Amos is robust software for SEM that allows users to validate their research and hypotheses by expanding conventional multivariate analysis techniques, such as regression, factor analysis, correlation, and analysis of variance. Using SPSS-Amos, one can develop intricate attitudinal and behavioral models that depict complex relationships more precisely than traditional multivariate statistical methods through an intuitive graphical or programmatic user interface (IBM, 2017; Dash et al., 2021).

SEM relies on path diagrams to provide a clear picture of the relationships between variables (Ullman and Bentler, 2012). For the SEM model fit, 100 sample size is qualified as fair; the current sample size is 258 beyond the threshold value (Ganbold et al., 2021). "Exploratory factor analysis (EFA)" and "confirmatory factor analysis (CFA)" were performed before the SEM analysis. EFA measures the correlation between constructs in a dataset by analyzing the correlations between them. It also reduces the broad data set to smaller set. Bartlett's test of sphericity and "Kaiser-Meyer-Olkin (KMO)" helps to understand the suitability of the data. The results show that the KMO value is 0.851(value>0.7), which is satisfactory. The measurement model is validated using CFA to identify how well the conceptual framework reflects the data. The results show that the Chi-square test result is 1.716 (< 3.0), the "Goodness fit index (GFI)" is 9.51 (>9.50), the "comparative fit index (CFI)" is 0.951, and the "Root mean square error of approximation (RMSEA)" is 0.053, which suggests that the model is a close fit.

## 5.1. Structural model assessment

To evaluate how reliable or consistent the questionnaires are, they are tested for dependability. The analysis reveals the extent to which measurement inaccuracies impact (or do not affect) the gathered data. In general, it is acknowledged that data can be regarded as credible if Cronbach's alpha is greater than 0.70 (Teo et al., 2015; Kumar et al. 2022). Table 3 shows that all the item loadings exceeded the 0.70 threshold, providing evidence of convergent validity (Jiang et al., 2020). Therefore, the items for each construct and the complete questionnaire have good internal consistency. Composite reliability (CR) in Table 4 shows that all six constructs are higher than 0.7, indicating that all the six are valid. Tabachnica and Fidell (2007) state that if the average variance extracted (AVE) is larger than 40%, the measurement questions represent the features of each study variable in the model. Each variable in Table 3 has an AVE value higher than 0.5, denoting convergent validity of the scale. Convergent validity can be further examined by analyzing the AVE t-values for factor loadings, and composite reliability (construct reliability) (Chau, 1997; Fornell & Larcker, 1981).

Discriminant validity can be evaluated by examining the square root of AVEs for each latent variable with its correlation to the other constructs (Fornell and Larcker, 1981). Table 4 shows the correlations between the constructs and demonstrates the discriminant validity of the variables. The correlation coefficient is higher than the square root of the AVE, and the AVE value for every construct is more than 0.5, suggesting that the constructs have significant discriminant coefficients, and that the model's intrinsic quality is ideal.

The path diagram of SEM prepared in "AMOS-20.0" is shown in Figure 2 and Figure A1 (Annexure). The results of the SEM and the path analysis of the conceptual framework are displayed in Table 5. They include "the path, coefficient, standard deviation, t-value, and p-value". The findings show that all nine hypotheses support, which leads to the conclusion that all the independent variables, COS, FSS, and TT, positively influence the dependent PEU, PU, and ITU. Statistical significance was found across all paths in the framework, with standardized "path coefficients ranging from 0.1 to 0.4". The path coefficient( $\beta$ ), (+/-1) indicates the degree of change in the outcome variable for each one unit change in the predictor variable. P values (<0.05) determine that the hypothesis results are statistically significant. The value shows a positive relationship between the independent variables (COS, FSS, and TT) and the dependent variables (PEU, PU, ITU). The findings reveal that food safety and security is perceived as the most important useful factor influencing the ITU the technology, followed by cost and traceability.



Figure 2: Measurement model (\*\*\* p<0.001, \*\* p<=0.01, \*p<= 0.05) (Authors own creation)

Constructs	Items	Loading factors	α	CR	AVE
	PU3	0.961			
Perceived	PU4	0.943			
usefulness	PU2	0.839	0.946	0.946	0.596
	PU5	0.811			
	PU1	0.804			
D 1	PEU3	0.861			
Perceived ease	PEU1	0.842	0.007	0.000	0.000
of use	PEU2	0.793	0.88/	0.888	0.602
	PEU4	0.757			
	TT1	0.952			
Transparency	TT5	0.923			
and traceability	TT2	0.730	0.899	0.9	0.539
	TT3	0.678			
	TT4	0.638			
	FSS3	0.994			
E. I. f. f. f 1	FSS2	0.935			
Food safety and	FSS5	0.719	0.911	0.913	0.543
security	FSS1	0.674			
	FSS4	0.673			
	COS3	0.798			
Cost	COS2	0.729	0.751	0.756	0.523
	COS4	0.618			
	ITU4	0.780			
	ITU3	0.778	]		
intention to use	ITU2	0.754	0.819	0.834	0.621
	ITU1	0.614	1		
	ITU5	0.584	1		

**Table 3:** Items; loading factors; Cronbach's alpha (α); Composite reliability (CR); average variance extracted (AVE) (Authors own creation)

Table 4:	Discriminant	t validity	(Authors	own creation)
			<b>\</b>	,

Fact	or	PU	FSS	TT	ITU	PEU	COS
PU		0.772					
FSS		0.53	0.776				
TT		0.393	0.348	0.734			
ITU		0.44	0.51	0.42	0.737		

PEU	0.343	0.311	0.381	0.327	0.723	
COS	0.14	0.221	0.214	0.173	0.297	0.788

"Hypot heses"	"Path"	"Coefficient"	"Standard error"	"t- statistics"	"P- values"	"Supported/ Not supported"
H1	COSPEU	0.284	0.1	2.837	**	Supported
H2	FSSPEU	0.2	0.081	2.471	*	Supported
H5	TTPEU	0.245	0.062	3.921	***	Supported
H3	FSSPU	0.409	0.063	6.437	***	Supported
H6	TTPU	0.15	0.046	3.269	***	Supported
H8	PEUPU	0.104	0.051	2.027	*	Supported
H9	PUITU	0.132	0.059	2.229	*	Supported
H4	FSSITU	0.248	0.059	4.192	***	Supported
H7	TT-ITU	0.126	0.039	3.204	***	Supported

**Table 5:** Path coefficients, standard error, t-statistics, and p-values (Authors own creation)

# Significant at P-value p < 0.05; \*\*\* p <= 0.001; \*\* p <= 0.01, \*p <= 0.05# t-statistics > 1;  $\beta$  strongest relationship (0 to 1)

## 5.2. Mediation effect

Mediation analysis in SEM is a statistical method to investigate the underlying mechanisms or pathways through which an independent variable affects a dependent variable. Mediation analysis helps identify one or more intermediate between the predictor and outcome variables and explain their relationship. The results of the mediation effect denote that all the paths (TT-PU-ITU, FSS-PU-ITU, TT-PEU-PU, FSS-PEU-PU) have a partial mediation effect. Partially mediated relationships assume that the mediating variable contributes to some of the relationships between the independent and dependent variables. Partially mediated relationships are not merely characterized by a substantial association between the mediator and the dependent variable but also by direct interactions between them. The results of the mediation effect are shown in Table 6.

Path	Direct effect	Indirect effect	Decision
TT—PU—ITU	0.245 (0.001)	0.041 (0.009)	PM
FSS—PUITU	0.374 (0.001)	0.079 (0.009)	PM
TT—PEU-PU	0.210 (0.001)	0.037 (0.003)	PM
FSS—PEU—PU	0.450 (0.001)	0.021 (0.008)	PM

**Table 6**: Results of mediation effect (Authors own creation)

\*\*Significant at P-value < 0.1; parenthesis values (P-values); Partial Mediation (PM)

#### 6. Discussion

The study identifies that the customer's willingness and the variables they perceived to be useful lead to adopting the technology. The study shows that FSS, COS, and TT positively influence ITU technology. Dwivedi et al. (2016) state that cost is essential in the implementation and intention to adopt the technology. The study hypothesis "H1( $\beta$  = 0.284, T= 2.837, p=0.005)" supports the claim that PEU, PU and ITU of the technology are influenced by cost. The path coefficient COS significantly relates to the predetermined variables PEU and PU. The study of (Wong et al., 2020a; Naval et al., 2021) agrees with our finding that cost saving significantly impacts the perceived usefulness and ITU of the technology. The analysis of the results indicates that the more the decisionmaker believes in the usefulness of the technology and the value for money it provides, the more the intention to adopt the technology increases. The  $\beta$  value of PEU and PU of the variable COS is less than FSS and TT as BLCT is a high-end technology, and implementation cost as an individual technology can be higher even if implementing it as a group will reduce its initial investment cost (Roeck et al. ,2020). It decreases overall expenses and supports more users using its services than a single user. Also, its safety mechanism is more secure as the users are verified under a certain criterion before it can be used. Customer preferences often lead to processing small volumes of transactions as quickly as possible and at a low cost, which can be possible by adopting BLCT. Chen et al. (2020) state that BLCT will help reduce the network's intermediaries, leading to cost reduction. Jiang et al. (2022) state that customers tend to emphasize whether the transactions can be approved quickly and cost-effectively. By investing in blockchain resources, all trading partners will be able to coordinate their activities on one platform, facilitating instant, transparent transactions that are free of delays (Ali et al., 2021). BLCT is believed to boost security while saving time and money by minimizing documentation processes (Karakas et al., 2021). However, the initial capital required to implement BLCT might cause difficulties for the developing AFSC sector and may limit its adoption.

FSS describes a food product's level of assurance that it will not cause illness or injury during its production, serving, or consumption, which helps customers build trust in the product by reducing their concerns about food products. The variable FSS and the supporting hypothesis "H2( $\beta$  = 0.2, T = 2.471, p = 0.013)"; "H3( $\beta$  = 0.409, T= 6.437, p<0.001)"; and "H4( $\beta$  = 0.248, T= 4.192, p<0.001)". state that FSS positively influences the intention and adoption of BLCT. A higher  $\beta$  value of the PEU and PU of FSS confirms that the FSS plays an important parameter that influences the adoption of the BLCT. The FSS, in particular, has not been discussed in any adoption model.

Nevertheless, Wang and Scrimgeour (2022) highlight that decision maker perception of food quality influences the intention to use the technology. The existing literature (Tan et al., 2018; Duan et al., 2020; Lin et al., 2020) provides a similar view. Integrating BLCT in the AFSC enhances decision confidence as it helps the food manufacturers to share information about the origin, batch numbers, and production dates, as well as promote food safety, certification, and organic products (Galvez et al., 2018), which act as a motivator for PEU and PU. BLCT enhances the ability to identify potential sources of contamination (Alladi et al., 2019), stop the illegal purchase of pesticides (Leng et al., 2018), and track the manufacturing and expiration date, which in turn helps reduce wastage. For instance, Walmart and Carrefour partnered with IBM's Food Trust blockchain network (Vu et al., 2021) to achieve real-time and end-to-end visibility at low costs.

Similarly, JBS integrated BLCT in their livestock farming to eliminate food fraud in their SC (Saha et al., 2022). These initiatives aim to enhance customers' trust and ensure safety and security in the AFSC. The decision-makers believe that a specific technology will help attain the objective, which leads to the ITU technology.

The variable TT of the SC can help increase decision-maker trust. Tracing the products to their origin has immensely helped to gain the trust of the decision-maker. Integrating BLCT in food traceability and transparency is an identification tool that helps track and trace food products quickly and accurately. The results "H5( $\beta$  = 0.245, T= 3.269, p<0.001)", "H6( $\beta$  = 0.15, T= 3.269, p=0.001)", and "H7( $\beta = 0.126$ , T= 3.204, p=0.001)" are in support of the hypotheses and show significant positive reliability to the adoption and intention to use the BLCT. The analysis supports (Queiroz and Wamba 2019; Wang et al., 2021; Wang and Scrimgeour, 2022), who state that TT is an essential determinant for the adoption of BLCT. TT brings trust and reliability to the system, which supports our analysis and is verified by (Queiroz et al., 2020) but in disagreement with (Queiroz et al., 2019). BLCT provides a reliable source of information about the origin of food products, allowing businesses to differentiate themselves from competitors (Vu. et al., 2021) and is perceived as valuable by decision-makers. However, the PU of TT falls short of FSS, maybe due to the SC members' lack of trust or the availability of other integrated technology like IoT, which is perceived as easier to use. Schulze-Ehlers et al. (2014) state that supplier, processor, and buyer asymmetry leads to a poor supplier-buyer relationship and an unoptimized SC. Li et al. (2022) and Wamba and Queiroz (2022) state that BLCT is ideal for increasing SC traceability, efficiencies, responsiveness, and managing supplier relationships. Adopting BLCT can help companies manage food safety crises more effectively, minimize the impact of food recalls, and enhance their SC productivity. BLCT's ability to track and trace also helps in storing and transporting the food, easing the process. In addition, it can also assist in speeding up the auditing and conflict resolution process, as financial records of all transactions can be documented and stored in the chain (Chang et al., 2020). Thus, the benefits of traceability and transparency are highly perceived to be useful and influence the adoption of BLCT in companies. In addition, the decision-maker will gain confidence in the provider and stop looking for alternative solutions, which will benefit the company (Villena et al., 2019). The technology's benefits contribute to perceived usefulness, significantly affecting the intention to use BLCT.

The last two Hypotheses (H8) and (H9) analysis show how PU and PEU of the technology positively influence the intention to use the technology. "H8( $\beta = 0.104$ , T= 2.027, p=0.043)" and "H9  $(\beta = 0.132, T = 2.229, p = 0.026)$ " support the hypotheses which are aligned with the findings in Lie et al. (2021) and Ullah et al. (2022). This suggests that the more decision-makers find the usefulness of the BLCT, the more they see the value of the technology and want to adopt it. Furthermore, the H9 has the smallest t value (t=2.027), which implies PEU of technology plays an important parameter in the ITU of the technology. The simpler the technology's ease of use, the better the decision-maker is ready to use it. Comparing our results with Queiroz et al. (2019), shows that trust, transparency, and traceability among the stakeholders and decision-makers influence the intention to adopt BLCT in the developing economy. To fully leverage the power of ease of use for BLCT adoption, organizations that use BLCT should showcase the benefits in their applications to increase customer trust. Governments and private organizations run many pilot projects to implement the technology in the system. In Latin America, the GrainChain startup company used BLCT to track the movement of the grain from farm to market, ensuring that farmers receive fair prices for their crops. Similarly, Agri10X, a startup in India, leverages digital technologies, including BLCT, to provide comprehensive solutions to every phase of the agricultural value chain, aiming to improve farmers' lives. By introducing trust in the system and lowering the cost of online transactions, BLCT also improves the organizations' synergy and efficiencies.

## 7. Conclusion

This research aims to shed light on the ITU BLCT in AFSC. According to the findings of this study, users intend to use a technology once they perceive its usefulness and ease of use. The study uses an empirical approach based on the proposed conceptual framework on TAM to analyze the ITU and the BLCT in the AFSC domain. The conceptual model was evaluated using SEM, and the results validated the proposed framework. Based on our statistical results, we can conclude that the perceived benefits of BLCT are translated into practical usefulness in three dimensions: food safety and security, transparency and traceability, and cost. According to the study, the perceived value of BLCT

among AFSC practitioners in India is not driven by hype but by the knowledge of the perceived benefits. The findings show that practitioners and decision-makers believe in the usefulness of BLCT, which enhances the SC by reducing food losses and wastage and increasing quality and efficiency. The decision-maker believes that integrating BLCT will help to trace and track the product and identify the origin and flow of the product and any potential sources of contamination occurring in the network. When SC intermediaries can get a good quality product and a visible SC, it helps to build trust between the supplier and consumer, leading to a higher likelihood of technology adoption.

Finally, it is essential to recognize that implementing BLCT cannot be expected to happen overnight, particularly in developing nations where infrastructure and regulatory frameworks are still evolving. However, instead of relying solely on government intervention or aid programs for progress, the global aid community should focus on establishing standards and enabling investors and companies to adopt diverse solutions. This approach will promote platform interoperability and provide potential investors with deeper insights into blockchain-based investment opportunities while emphasizing their societal benefits within the AFSC domain. A well-designed standardization system could enhance transparency across borders by streamlining cross-border transactions and promoting ethical practices among businesses operating in emerging markets- ultimately benefiting both stakeholders involved and facilitating sustainable economic growth globally.

## 7.1. Research Contributions

# 7.1.1. Theoretical Contribution

The current study contributes to both theory and practice. Our research studies the acceptance of blockchain adoption technology by providing a TAM approach practical for innovative technologybased services. We find several factors that positively influence AFSC's intention to use BLCT. Our study highlights the driving factors, FSS, TT, and COS, influencing the intention to use BLCT based on TAM. Our results highlight the importance of including these factors as they help build the lane for adopting the technology.

As theorized by TAM, both PU and PEU significantly influence the intention to use BLCT (Yen et al., 2010). According to the empirical findings of (Kamble et al., 2018), PU is influenced by PEU, and the PU is explained by 28%, suggesting other factors may be at play. Our study shows similar results:  $R^2$  (PEU)= 0.37,  $R^2$  (PU)= 0.21 with three different constructs for PU. The value of  $R^2$  accounted for PU for (COS)= 0.24, (TT)= 0.65, and (FSS)=0.38. The study implies that the higher the perceived benefits of implementing BLCT (saving time, efficiency, and convenience), the more likely users are to use it. The empirical results also indicate that BLCT's characteristics can influence

PU and PEU directly; the higher the ease of use and perceived benefit, the higher the chances of intention to use.

Furthermore, these findings highlight the importance of addressing technical considerations and user perceptions when introducing new technology solutions into an organization's workflow. This approach will help foster greater employee acceptance and engagement while maximizing ROI. The contingency perspective in this study advances our understanding of technology acceptance and makes the results relevant to practice.

## 7.1.2. Managerial Contribution

The emergence of BLCT technology has been a significant development in agriculture and food security. However, as this technology is still in its inception stage, it requires a more comprehensive understanding and awareness among manufacturers, policymakers, governments, and individuals for its effective implementation on a wider scale. BLCT demands higher technical knowledge from system designers or developers than traditional agricultural technologies, so they must simplify the setup process by providing clear instructions with an accompanying manual. Additionally, managers should collaborate closely with technology service providers to mitigate the high computing costs of adopting this new system. In particular, initial investment costs may be prohibitively expensive for some companies, which could deter them from investing in such advanced systems; therefore, reducing these expenses would encourage further adoption of this innovative approach toward farming practices - particularly given the increased demand for skilled professionals required during installation phases. Furthermore, given the potential benefits of BLCT technology for food security and agricultural sustainability, policymakers must develop supportive regulatory frameworks to promote investment in BLCT research and development.

#### 7.2 Limitations and future research direction

Although the study provides insight into the intentions of using BLCT in AFSC, further research is required to address some limitations. Firstly, for the empirical study to find the ITU BLCT in AFSC, only the single SC decision maker perspective has been observed. Future research focusing on the dual perspective of the supplier- buyer's ITU the technology is recommended. Secondly, while transparency and traceability, food safety and security, and cost were deemed essential variables examined by this study, incorporating additional factors such as technical knowledge, government intervention, regulatory framework about governance, and policy implementation practices aimed at societal benefits are recommended avenues worthy of exploration. Thirdly, the model is tested on a sample of 258 working professionals in the Indian AFSC domain, thus restricting its generalizability across other countries unless supplemented by further studies.

Nevertheless, practitioners, researchers, and scholars can apply and extend our proven model to other nations through additional research. A comparative study between the developed and developing nations can be conducted to understand ITU BLCT. In addition, the food- and agriculture-specific industries and country-specific ones may be examined to understand better the variables affecting or leading to adopting BLCT. Furthermore, a longitudinal study of practitioners' perceptions and the advanced industrial use cases would provide great insight into time-tested results.

## References

- Agrawal, P. and Narain, R. (2023), "Analysis of enablers for the digitalization of supply chain using an interpretive structural modelling approach", International Journal of Productivity and Performance Management, Vol. 72 No. 2, pp. 410-439. https://doi.org/10.1108/IJPPM-09-2020-0481
- Ahmed, W.A.H., MacCarthy, B.L. and Treiblmaier, H. (2022), "Why, where and how are organizations using blockchain in their supply chains? Motivations, application areas and contingency factors", *International Journal of Operations & Production Management*, Vol. 42 No. 12, pp. 1995–2028, doi: 10.1108/IJOPM-12-2021-0805.
- Ali, I. and Aboelmaged, M.G.S. (2022), "Implementation of supply chain 4.0 in the food and beverage industry: perceived drivers and barriers", International Journal of Productivity and Performance Management, Vol. 71 No. 4, pp. 1426–1443, doi: 10.1108/IJPPM-07-2020-0393
- Ali, I. and Govindan, K. (2023), "Extenuating operational risks through digital transformation of agri-food supply chains", *Production Planning & Control*, Vol. 34 No. 12, pp. 1165–1177, doi: 10.1080/09537287.2021.1988177.
- Ali, M.H., Chung, L., Kumar, A., Zailani, S. and Tan, K.H. (2021), "A sustainable Blockchain framework for the halal food supply chain: Lessons from Malaysia", Technological Forecasting and Social Change, Vol. 170, p. 120870, doi: 10.1016/j.techfore.2021.120870.
- Alladi, T., Chamola, V., Parizi, R. M., & Choo, K. K. R. (2019). Blockchain applications for industry 4.0 and industrial IoT: A review. *IEEE Access*, 7, pp. 176935-176951.
- Amoako-Gyampah, K. (2007), "Perceived usefulness, user involvement and behavioral intention: An empirical study of ERP implementation", *Computers in Human Behavior*, Vol. 23, No. 3, pp. 1232–1248. https://doi.org/10.1016/j.chb.2004.12.002

- Antonucci, F., Figorilli, S., Costa, C., Pallottino, F., Raso, L., & Menesatti, P. (2019), "A Review on blockchain applications in the agri-food sector", *Journal of the Science of Food and Agriculture*, Vol. 99, No. 14, pp. 6129-6138, doi: 10.1002/jsfa.9912.
- Armstrong, J. S., & Overton, T. S. (1977). Estimating Nonresponse Bias in Mail Surveys. *Journal of Marketing Research*, SAGE Publications, 14(3), 396–402.
- Asokan, D. R., Huq, F. A., Smith, C. M., & Stevenson, M. (2022), "Socially responsible operations in the Industry 4.0 era: post-COVID-19 technology adoption and perspectives on future research", *International Journal of Operations & Production Management*, Vol. 42 No. 13, pp. 185-217. <u>https://doi.org/10.1108/IJOPM-01-2022-0069</u>
- Astill, J., Dara, R.A., Campbell, M., Farber, J.M., Fraser, E.D.G., Sharif, S. and Yada, R.Y. (2019),
  "Transparency in food supply chains: A review of enabling technology solutions", *Trends in Food Science & Technology*, Vol. 91, pp. 240–247, doi: 10.1016/j.tifs.2019.07.024.
- Baharmand, H., Maghsoudi, A. and Coppi, G. (2021), "Exploring the application of blockchain to humanitarian supply chains: insights from Humanitarian Supply Blockchain pilot project", *International Journal of Operations & Production Management*, Vol. 41 No. 9, pp. 1522–1543, doi: 10.1108/IJOPM-12-2020-0884.
- Barbosa, M.W. (2021), "Uncovering research streams on agri-food supply chain management: A bibliometric study", *Global Food Security*, Vol. 28, p. 100517, doi: 10.1016/j.gfs.2021.100517.
- Cao, S., Powell, W., Foth, M., Natanelov, V., Miller, T. and Dulleck, U. (2021), "Strengthening consumer trust in beef supply chain traceability with a blockchain-based human-machine reconcile mechanism", Computers and Electronics in Agriculture, Vol. 180, p. 105886, doi: 10.1016/j.compag.2020.105886.
- Chang, S.E. and Chen, Y. (2020), "When Blockchain Meets Supply Chain: A Systematic Literature Review on Current Development and Potential Applications", IEEE Access, Vol. 8, pp. 62478–62494, doi: 10.1109/ACCESS.2020.2983601.
- Chang, Y., Iakovou, E. and Shi, W. (2020), "Blockchain in global supply chains and cross border trade: a critical synthesis of the state-of-the-art, challenges and opportunities", International Journal of Production Research, Vol. 58 No. 7, pp. 2082–2099, doi: 10.1080/00207543.2019.1651946.

- Chavalala, M.M., Bag, S., Pretorius, J.H.C. and Rahman, M.S. (2022), "A multi-method study on the barriers of the blockchain technology application in the cold supply chains", Journal of Enterprise Information Management, doi: 10.1108/JEIM-06-2022-0209.
- Chau, P. Y. (1997). Re-examining a model for evaluating information center success using a structural equation modelling approach. *Decision Sciences*, Vol. 28, No. 2, pp. 309-334.
- Chen, S., Liu, X., Yan, J., Hu, G. and Shi, Y. (2021), "Processes, benefits, and challenges for adoption of blockchain technologies in food supply chains: a thematic analysis", Information Systems and E-Business Management, Vol. 19 No. 3, pp. 909–935, doi: 10.1007/s10257-020-00467-3.
- Chen, Y., Richter, J.I. and Patel, P.C. (2021), "Decentralized Governance of Digital Platforms", Journal of Management, Vol. 47 No. 5, pp. 1305–1337, doi: 10.1177/0149206320916755.
- Dash, G. and Paul, J. (2021), "CB-SEM vs PLS-SEM methods for research in social sciences and technology forecasting", Technological Forecasting and Social Change, Vol. 173, p. 121092, doi: 10.1016/j.techfore.2021.121092.
- Davis, F. D. (1985). A technology acceptance model for empirically testing new end-user information systems: Theory and results (Doctoral dissertation, Massachusetts Institute of Technology).
- Davis, F.D. (1989), "Perceived Usefulness, Perceived Ease of Use, and User Acceptance of Information Technology", MIS Quarterly, Vol. 13 No. 3, p. 319, doi: 10.2307/249008.
- Danese, P., Mocellin, R., & Romano, P. (2021). Designing blockchain systems to prevent counterfeiting in wine supply chains: a multiple-case study. *International Journal of Operations & Production Management*, Vol. 41, No. 13, pp. 1-33. <u>https://doi.org/10.1108/IJOPM-12-2019-0781</u>
- Duan, J., Zhang, C., Gong, Y., Brown, S. and Li, Z. (2020), "A Content-Analysis Based Literature Review in Blockchain Adoption within Food Supply Chain", International Journal of Environmental Research and Public Health, Vol. 17 No. 5, p. 1784, doi: 10.3390/ijerph17051784.
- Du, M., Chen, Q., Xiao, J., Yang, H., & Ma, X. (2020), "Supply chain finance innovation using blockchain", *IEEE Transactions on Engineering Management*, Vol. 67, No. 4, pp. 1045-1058. Doi: 10.1109/TEM.2020.2971858.
- Dwivedi, Y.K., Rana, N.P., Jeyaraj, A., Clement, M. and Williams, M.D. (2019), "Re-examining the Unified Theory of Acceptance and Use of Technology (UTAUT): Towards a Revised

Theoretical Model", Information Systems Frontiers, Vol. 21 No. 3, pp. 719–734, doi: 10.1007/s10796-017-9774-y.

- Dwivedi, Y.K., Shareef, M.A., Simintiras, A.C., Lal, B. and Weerakkody, V. (2016), "A generalised adoption model for services: A cross-country comparison of mobile health (mhealth)", Government Information Quarterly, Vol. 33 No. 1, pp. 174–187, doi: 10.1016/j.giq.2015.06.003.
- FAO 2009, Global agriculture towards 2050 Food and Agriculture Organization https://www.fao.org/fileadmin/templates/wsfs/docs/Issues\_papers/HLEF2050\_Global\_Agric ulture.pdf (Assessed date 15-01-2023)
- Farshidi, S., Jansen, S., España, S., & Verkleij, J. (2020), "Decision support for blockchain platform selection: Three industry case studies", *IEEE Transactions on Engineering Management*, Vol. 67, No. 4, pp. 1109-1128. Doi: 10.1109/TEM.2019.2956897.
- Feng, H., Wang, X., Duan, Y., Zhang, J. and Zhang, X. (2020), "Applying blockchain technology to improve agri-food traceability: A review of development methods, benefits and challenges", *Journal of Cleaner Production*, Vol. 260, p. 121031, doi: 10.1016/j.jclepro.2020.121031.
- Fischer, C. (2013), "Trust and communication in European agri-food chains", Supply Chain Management: An International Journal, Vol.18, No. 2, pp. 208-218. https://doi.org/10.1108/13598541311318836
- Fornell, C. and Larcker, D.F. (1981), "Structural Equation Models with Unobservable Variables and Measurement Error: Algebra and Statistics", *Journal of Marketing Research*, Vol. 18 No. 3, p. 382, doi: 10.2307/3150980.
- Ganbold, O., Matsui, Y. and Rotaru, K. (2021), "Effect of information technology-enabled supply chain integration on firm's operational performance", *Journal of Enterprise Information Management*, Vol. 34 No. 3, pp. 948–989, doi: 10.1108/JEIM-10-2019-0332.
- Galvez, J.F., Mejuto, J.C. and Simal-Gandara, J. (2018), "Future challenges on the use of blockchain for food traceability analysis", *TrAC Trends in Analytical Chemistry*, Vol. 107, pp. 222–232, doi: 10.1016/j.trac.2018.08.011.
- Gardas, B.B., Raut, R.D. and Narkhede, B. (2017), "Modeling causal factors of post-harvesting losses in vegetable and fruit supply chain: An Indian perspective", Renewable and Sustainable Energy Reviews, Vol. 80, pp. 1355–1371, doi: 10.1016/j.rser.2017.05.259.

- Gardas, B., Raut, R., Jagtap, A. H., & Narkhede, B. (2019), "Exploring the key performance indicators of green supply chain management in agro-industry", *Journal of Modelling in Management, Vol.* 14, No. 1, pp. 260–283. https://doi.org/10.1108/JM2-12-2017-0139
- Guggenberger, T., Schweizer, A., & Urbach, N. (2020). Improving interorganizational information sharing for vendor managed inventory: Toward a decentralized information hub using blockchain technology. *IEEE Transactions on Engineering Management*, Vol 67, No. 4, pp. 1074-1085. Doi: 10.1109/TEM.2020.2978628.
- Gupta, R., & Shankar, R. (2023), "Managing food security using blockchain-enabled traceability system", *Benchmarking: An International Journal*. <u>https://doi.org/10.1108/BIJ-01-2022-0029</u>
- Hassani, H., Huang, X. and Silva, E. (2018), "Banking with blockchain-ed big data", Journal of Management Analytics, Vol. 5 No. 4, pp. 256–275, doi: 10.1080/23270012.2018.1528900.
- Hew, J.-J., Wong, L.-W., Tan, G.W.-H., Ooi, K.-B. and Lin, B. (2020), "The blockchain-based Halal traceability systems: a hype or reality?", Supply Chain Management: An International Journal, Vol. 25 No. 6, pp. 863–879, doi: 10.1108/SCM-01-2020-0044.
- Iftekhar, A., Cui, X., Hassan, M., & Afzal, W. (2020), "Application of blockchain and Internet of Things to ensure tamper-proof data availability for food safety", *Journal of Food Quality*, Vol. 2020. https://doi.org/10.1155/2020/5385207
- (IBM, 2017), SPSS Amos | IBM". (n.d.)., available at: <u>https://www.ibm.com/products/structural-equation-modeling-sem</u> (accessed 12 February 2024).
- Jiang, X., Wang, G., De Clercq, D., & Yi, X. (2020), "How Do Firms Achieve Successful Technology Commercialization? Evidence from Chinese Manufacturing Firms", *IEEE Transactions on Engineering Management*. Vol. 69, No. 5, pp. 1973-1986. Doi: 10.1109/TEM.2020.2997965.
- Jiang, S., Li, Y., Wang, S. and Zhao, L. (2022), "Blockchain competition: The tradeoff between platform stability and efficiency", European Journal of Operational Research, Vol. 296 No. 3, pp. 1084–1097, doi: 10.1016/j.ejor.2021.05.031.
- Kafetzopoulos, D., Margariti, S., Stylios, C., Arvaniti, E., & Kafetzopoulos, P. (2023), "Managing the traceability system for food supply chain performance", *International Journal of Productivity and Performance Management*. <u>https://doi.org/10.1108/IJPPM-12-2021-0690</u>
- Kamath, R. (2018), "Food Traceability on Blockchain: Walmart's Pork and Mango Pilots with IBM", The Journal of the British Blockchain Association, Vol. 1 No. 1, pp. 1–12, doi: 10.31585/jbba-1-1-(10)2018.

- Kamble, S., Gunasekaran, A. and Arha, H. (2019), "Understanding the Blockchain technology adoption in supply chains-Indian context", International Journal of Production Research, Vol. 57 No. 7, pp. 2009–2033, doi: 10.1080/00207543.2018.1518610.
- Kamilaris, A., Fonts, A. and Prenafeta-Boldú, F.X. (2019), "The rise of blockchain technology in agriculture and food supply chains", Trends in Food Science & Technology, Vol. 91, pp. 640– 652, doi: 10.1016/j.tifs.2019.07.034.
- Karamchandani, A., Srivastava, S.K., Kumar, S. and Srivastava, A. (2021), "Analysing perceived role of blockchain technology in SCM context for the manufacturing industry", International Journal of Production Research, Vol. 59 No. 11, pp. 3398–3429, doi: 10.1080/00207543.2021.1883761.
- Karakas, S., Acar, A.Z. and Kucukaltan, B. (2021), "Blockchain adoption in logistics and supply chain: a literature review and research agenda", International Journal of Production Research, pp. 1–24, doi: 10.1080/00207543.2021.2012613.
- Kayikci, Y., Subramanian, N., Dora, M. and Bhatia, M.S. (2022), "Food supply chain in the era of Industry 4.0: blockchain technology implementation opportunities and impediments from the perspective of people, process, performance, and technology", Production Planning & Control, Vol. 33 No. 2–3, pp. 301–321, doi: 10.1080/09537287.2020.1810757.
- Khan, S., Kaushik, M.K., Kumar, R. and Khan, W. (2023), "Investigating the barriers of blockchain technology integrated food supply chain: a BWM approach", Benchmarking: An International Journal, Vol. 30 No. 3, pp. 713–735, doi: 10.1108/BIJ-08-2021-0489.
- Ko, T., Lee, J. and Ryu, D. (2018), "Blockchain Technology and Manufacturing Industry: Real-Time Transparency and Cost Savings", Sustainability, Vol. 10 No. 11, p. 4274, doi: 10.3390/su10114274.
- Krzyzanowski Guerra, K. and Boys, K.A. (2022), "A new food chain: Adoption and policy implications to blockchain use in agri-food industries", Applied Economic Perspectives and Policy, Vol. 44 No. 1, pp. 324–349, doi: 10.1002/aepp.13163.
- Kshetri, N. (2019), "Blockchain and the Economics of Food Safety", IT Professional, Vol. 21 No. 3, pp. 63–66, doi: 10.1109/MITP.2019.2906761.
- Kumar, S., Raut, R.D., Priyadarshinee, P., Mangla, S.K., Awan, U. and Narkhede, B.E. (2022), "The Impact of IoT on the Performance of Vaccine Supply Chain Distribution in the COVID-19 Context", IEEE Transactions on Engineering Management, pp. 1–11, doi: 10.1109/TEM.2022.3157625.

- Kumar, M., Choubey, V.K., Raut, R.D. and Jagtap, S. (2023a), "Enablers to achieve zero hunger through IoT and blockchain technology and transform the green food supply chain systems", Journal of Cleaner Production, Vol. 405, p. 136894, doi: 10.1016/j.jclepro.2023.136894.
- Kumar, M., Raut, R.D., Mangla, S.K., Chowdhury, S. and Choubey, V.K. (2024), "Moderating ESG compliance between industry 4.0 and green practices with green servitization: Examining its impact on green supply chain performance", Technovation, Vol. 129, p. 102898, doi: 10.1016/j.technovation.2023.102898.
- Kumar, V., Yetkin Ekren, B., Wang, J., Shah, B., & Frederico, G. F. (2023), "Investigating the impact of COVID-19 on sustainable food supply chains", *Journal of Modelling in Management*, Vol.18, No.4, pp. 1250–1273. https://doi.org/10.1108/JM2-03-2022-0072
- Lakhani, K. R., & Iansiti, M. (2017), "The truth about blockchain", *Harvard Business Review*, Vol. 95, No.1, pp. 119-127.
- Lanzolla, G. and Suarez, F.F. (2012), "Closing the Technology Adoption–Use Divide: The Role of Contiguous User Bandwagon", Journal of Management, Vol. 38 No. 3, pp. 836–859, doi: 10.1177/0149206310369938.
- Li, G., Xue, J., Li, N. and Ivanov, D. (2022), "Blockchain-supported business model design, supply chain resilience, and firm performance", Transportation Research Part E: Logistics and Transportation Review, Vol. 163, p. 102773, doi: 10.1016/j.tre.2022.102773.
- Li, K., Lee, J.-Y. and Gharehgozli, A. (2023), "Blockchain in food supply chains: a literature review and synthesis analysis of platforms, benefits and challenges", International Journal of Production Research, Vol. 61 No. 11, pp. 3527–3546, doi: 10.1080/00207543.2021.1970849.
- Liu, W., Shao, X.-F., Wu, C.-H. and Qiao, P. (2021), "A systematic literature review on applications of information and communication technologies and blockchain technologies for precision agriculture development", Journal of Cleaner Production, Vol. 298, p. 126763, doi: 10.1016/j.jclepro.2021.126763.
- Liu, W., Wang, J., Jia, F. and Choi, T.-M. (2022), "Blockchain announcements and stock value: a technology management perspective", International Journal of Operations & Production Management, Vol. 42 No. 5, pp. 713–742, doi: 10.1108/IJOPM-08-2021-0534.
- Luarn, P. and Lin, H.-H. (2005), "Toward an understanding of the behavioral intention to use mobile banking", Computers in Human Behavior, Vol. 21 No. 6, pp. 873–891, doi: 10.1016/j.chb.2004.03.003.

- Mangla, S.K., Kazancoglu, Y., Ekinci, E., Liu, M., Özbiltekin, M. and Sezer, M.D. (2021), "Using system dynamics to analyze the societal impacts of blockchain technology in milk supply chainsrefer", Transportation Research Part E: Logistics and Transportation Review, Vol. 149, p. 102289, doi: 10.1016/j.tre.2021.102289.
- Marucheck, A., Greis, N., Mena, C. and Cai, L. (2011), "Product safety and security in the global supply chain: Issues, challenges and research opportunities", Journal of Operations Management, Vol. 29 No. 7–8, pp. 707–720, doi: 10.1016/j.jom.2011.06.007.
- Masudin, I., Ramadhani, A., Restuputri, D.P. and Amallynda, I. (2021), "The Effect of Traceability System and Managerial Initiative on Indonesian Food Cold Chain Performance: A Covid-19 Pandemic Perspective", Global Journal of Flexible Systems Management, Vol. 22 No. 4, pp. 331–356, doi: 10.1007/s40171-021-00281-x.
- Mollenkopf, D.A., Peinkofer, S.T. and Chu, Y.( J. (2022), "Supply chain transparency: Consumer reactions to incongruent signals", Journal of Operations Management, Vol. 68 No. 4, pp. 306– 327, doi: 10.1002/joom.1180.
- Mondal, S., Wijewardena, K.P., Karuppuswami, S., Kriti, N., Kumar, D. and Chahal, P. (2019),
  "Blockchain Inspired RFID-Based Information Architecture for Food Supply Chain", IEEE
  Internet of Things Journal, Vol. 6 No. 3, pp. 5803–5813, doi: 10.1109/JIOT.2019.2907658.
- Mukherjee, A.A., Singh, R.K., Mishra, R. and Bag, S. (2022), "Application of blockchain technology for sustainability development in agricultural supply chain: justification framework", Operations Management Research, Vol. 15 No. 1–2, pp. 46–61, doi: 10.1007/s12063-021-00180-5.
- Mukri, B. (2018), "Blockchain technology in supply chain management: a review", *International Research Journal of Engineering and Technology*, Vol. 5, No. (6), pp. 2497-2500.
- Nayal, K., Raut, R.D., Yadav, V.S., Priyadarshinee, P. and Narkhede, B.E. (2022), "The impact of sustainable development strategy on sustainable supply chain firm performance in the digital transformation era", Business Strategy and the Environment, Vol. 31 No. 3, pp. 845–859, doi: 10.1002/bse.2921.
- Nayal, K., Raut, R.D., Narkhede, B.E., Priyadarshinee, P., Panchal, G.B. and Gedam, V.V. (2023), "Antecedents for blockchain technology-enabled sustainable agriculture supply chain", Annals of Operations Research, Vol. 327 No. 1, pp. 293–337, doi: 10.1007/s10479-021-04423-3.

- Paul, T., Mondal, S., Islam, N. and Rakshit, S. (2021), "The impact of blockchain technology on the tea supply chain and its sustainable performance", Technological Forecasting and Social Change, Vol. 173, p. 121163, doi: 10.1016/j.techfore.2021.121163.
- Podsakoff, P.M., MacKenzie, S.B., Lee, J.-Y. and Podsakoff, N.P. (2003), "Common method biases in behavioral research: A critical review of the literature and recommended remedies.", Journal of Applied Psychology, Vol. 88 No. 5, pp. 879–903, doi: 10.1037/0021-9010.88.5.879.
- Podsakoff, P.M. and Organ, D.W. (1986), "Self-Reports in Organizational Research: Problems and Prospects", Journal of Management, Vol. 12 No. 4, pp. 531–544, doi: 10.1177/014920638601200408.
- Queiroz, M.M. and Fosso Wamba, S. (2019), "Blockchain adoption challenges in supply chain: An empirical investigation of the main drivers in India and the USA", International Journal of Information Management, Vol. 46, pp. 70–82, doi: 10.1016/j.ijinfomgt.2018.11.021.
- Queiroz, M.M., Fosso Wamba, S., De Bourmont, M. and Telles, R. (2021), "Blockchain adoption in operations and supply chain management: empirical evidence from an emerging economy", International Journal of Production Research, Vol. 59 No. 20, pp. 6087–6103, doi: 10.1080/00207543.2020.1803511.
- Rana, N.P., Dwivedi, Y.K., Lal, B., Williams, M.D. and Clement, M. (2017), "Citizens' adoption of an electronic government system: towards a unified view", Information Systems Frontiers, Vol. 19 No. 3, pp. 549–568, doi: 10.1007/s10796-015-9613-y.
- Rana, N.P., Dwivedi, Y.K., Williams, M.D. and Weerakkody, V. (2016), "Adoption of online public grievance redressal system in India: Toward developing a unified view", Computers in Human Behavior, Vol. 59, pp. 265–282, doi: 10.1016/j.chb.2016.02.019.
- Roeck, D., Sternberg, H. and Hofmann, E. (2020), "Distributed ledger technology in supply chains: a transaction cost perspective", International Journal of Production Research, Vol. 58 No. 7, pp. 2124–2141, doi: 10.1080/00207543.2019.1657247.
- Rogerson, M. and Parry, G.C. (2020), "Blockchain: case studies in food supply chain visibility", Supply Chain Management: An International Journal, Vol. 25 No. 5, pp. 601–614, doi: 10.1108/SCM-08-2019-0300.
- Ronaghi, M.H. (2021), "A blockchain maturity model in agricultural supply chain", Information Processing in Agriculture, Vol. 8 No. 3, pp. 398–408, doi: 10.1016/j.inpa.2020.10.004.

- Roy, V. and Srivastava, S.K. (2022), "The safety-quality dominant view of food chain integrity: Implications for consumer-centric food chain governance", International Journal of Management Reviews, Vol. 24 No. 1, pp. 3–24, doi: 10.1111/ijmr.12258.
- Saha, A., Raut, R. and Kumar, M. (2023), Digital technology adoption challenges in the agri-food supply chain from the perspective of attaining sustainable development goals, *The International Journal of Logistics Management*. <u>https://doi.org/10.1108/IJLM-09-2023-0412</u>
- Saha, A., Raut, R.D. and Kumar, M. (2023a), Leveraging Blockchain Technology to Combat Food Fraud in the Agri-food Supply Chain. Int J Food Sci Technol. Accepted Author Manuscript. https://doi.org/10.1111/ijfs.16889
- Salah, K., Nizamuddin, N., Jayaraman, R. and Omar, M. (2019), "Blockchain-Based Soybean Traceability in Agricultural Supply Chain", IEEE Access, Vol. 7, pp. 73295–73305, doi: 10.1109/ACCESS.2019.2918000.
- Saurabh, K., Rani, N. and Upadhyay, P. (2023), "Towards blockchain led decentralized autonomous organization (DAO) business model innovations", Benchmarking: An International Journal, Vol. 30 No. 2, pp. 475–502, doi: 10.1108/BIJ-10-2021-0606.
- Saurabh, S. and Dey, K. (2021), "Blockchain technology adoption, architecture, and sustainable agrifood supply chains", Journal of Cleaner Production, Vol. 284, p. 124731, doi: 10.1016/j.jclepro.2020.124731.
- Schmidt, C.G. and Wagner, S.M. (2019), "Blockchain and supply chain relations: A transaction cost theory perspective", Journal of Purchasing and Supply Management, Vol. 25 No. 4, p. 100552, doi: 10.1016/j.pursup.2019.100552.
- Schulze-Ehlers, B., Steffen, N., Busch, G. and Spiller, A. (2014), "Supply chain orientation in SMEs as an attitudinal construct: Conceptual considerations and empirical application to the dairy sector", Supply Chain Management: An International Journal, Vol. 19 No. 4, pp. 395–412, doi: 10.1108/SCM-07-2013-0241.
- Shah, R. and Goldstein, S.M. (2006), "Use of structural equation modeling in operations management research: Looking back and forward", Journal of Operations Management, Vol. 24 No. 2, pp. 148–169, doi: 10.1016/j.jom.2005.05.001.
- Sharma, M. G. (2023), "Supply chain, geographical indicator and blockchain: Provenance model for commodity", *International Journal of Productivity and Performance Management*, Vol 72, No. 1, pp. 92–108. <u>https://doi.org/10.1108/IJPPM-05-2021-0288</u>

- Shi, J. and Jiang, Z. (2023), "Willingness to pay a premium price for green products: does a reference group matter?", Environment, Development and Sustainability, Vol. 25 No. 8, pp. 8699– 8727, doi: 10.1007/s10668-022-02419-y.
- Srivastava, A. and Dashora, K. (2022), "Application of blockchain technology for agrifood supply chain management: a systematic literature review on benefits and challenges", Benchmarking: An International Journal, Vol. 29 No. 10, pp. 3426–3442, doi: 10.1108/BIJ-08-2021-0495.
- Sternberg, H.S., Hofmann, E. and Roeck, D. (2021), "The Struggle is Real: Insights from a Supply Chain Blockchain Case", Journal of Business Logistics, Vol. 42 No. 1, pp. 71–87, doi: 10.1111/jbl.12240.
- Susanty, A., Puspitasari, N. B., Prastawa, H., & Renaldi, S. V. (2021), "Exploring the best policy scenario plan for the dairy supply chain: A DEMATEL approach", *Journal of Modelling in Management*, Vol. 16, No. 1, pp. 240–266. <u>https://doi.org/10.1108/JM2-08-2019-0185</u>
- Sunny, J., Undralla, N. and Madhusudanan Pillai, V. (2020), "Supply chain transparency through blockchain-based traceability: An overview with demonstration", Computers & Industrial Engineering, Vol. 150, p. 106895, doi: 10.1016/j.cie.2020.106895.
- Tabachnick, B. G., Fidell, L. S., & Ullman, J. B. (2007), "Using Multivariate Statistics" (Vol. 5, pp. 481-498). Boston, MA: Pearson.
- Teo, A. C., Tan, G. W. H., Ooi, K. B., Hew, T. S., & Yew, K. T. (2015), "The effects of convenience and speed in m-payment", *Industrial Management & Data Systems*, Vol. 115 No. 2, pp. 311-331).
- Tian, F. (2016) An Agri-Food Supply Chain Traceability System for China Based on RFID & Blockchain Technology 13th International Conference on Service Systems and Service Management. Kunming, China, 24-26 June 2016 Beijing: China.
- Tran, L.T.T. and Nguyen, P.T. (2021), "Co-creating blockchain adoption: theory, practice and impact on usage behavior", Asia Pacific Journal of Marketing and Logistics, Vol. 33 No. 7, pp. 1667– 1684, doi: 10.1108/APJML-08-2020-0609.
- Treiblmaier, H. (2018), "The Impact of the Blockchain on the Supply Chain: A Theory-Based Research Framework and a Call for Action", SSRN Electronic Journal, doi: 10.2139/ssrn.3224145.
- Trollman, H., Garcia-Garcia, G., Jagtap, S., & Trollman, F. (2022), "Blockchain for ecologically embedded coffee supply chains", Logistics, Vol. 6, No. 3, pp. 43. https://doi.org/10.3390/logistics6030043

- Ullman, J. B., & Bentler, P. M. (2012). Structural equation modelling. In Handbook of Psychology (2nd ed. Vol. 2).
- Vasan, M. and Yoganandan, G. (2023), "Does the belief of farmers on land as God influence the adoption of smart farming technologies?", Benchmarking: An International Journal, doi: 10.1108/BIJ-10-2022-0645.
- Venkatesh, V. (2022), "Adoption and use of AI tools: a research agenda grounded in UTAUT", Annals of Operations Research, Vol. 308 No. 1–2, pp. 641–652, doi: 10.1007/s10479-020-03918-9.
- Venkatesh, V. and Bala, H. (2008), "Technology Acceptance Model 3 and a Research Agenda on Interventions", Decision Sciences, Vol. 39 No. 2, pp. 273–315, doi: 10.1111/j.1540-5915.2008.00192.x.
- Venkatesh, V. and Davis, F.D. (2000), "A Theoretical Extension of the Technology Acceptance Model: Four Longitudinal Field Studies", Management Science, Vol. 46 No. 2, pp. 186–204, doi: 10.1287/mnsc.46.2.186.11926.
- Venkatesh, Morris, Davis, and Davis. (2003), "User Acceptance of Information Technology: Toward a Unified View", MIS Quarterly, Vol. 27 No. 3, p. 425, doi: 10.2307/30036540.
- Venkatesh, Thong, and Xu. (2012), "Consumer Acceptance and Use of Information Technology: Extending the Unified Theory of Acceptance and Use of Technology", MIS Quarterly, Vol. 36 No. 1, p. 157, doi: 10.2307/41410412.
- Vikaliana, R., Rasi, R.Z.R.M. and Pujawan, I.N. (2021), "Traceability System on Mangosteen Supply Chain Management Using Blockchain Technology: A Model Design", Studies of Applied Economics, Vol. 39 No. 4, doi: 10.25115/eea.v39i4.4565.
- Villena, V.H., Choi, T.Y. and Revilla, E. (2019), "Revisiting Interorganizational Trust: Is More Always Better or Could More Be Worse?", Journal of Management, Vol. 45 No. 2, pp. 752– 785, doi: 10.1177/0149206316680031.
- Vu, N., Ghadge, A. and Bourlakis, M. (2023), "Blockchain adoption in food supply chains: a review and implementation framework", Production Planning & Control, Vol. 34 No. 6, pp. 506– 523, doi: 10.1080/09537287.2021.1939902.
- Wamba, S.F. and Queiroz, M.M. (2019), "The Role of Social Influence in Blockchain Adoption: The Brazilian Supply Chain Case", IFAC-PapersOnLine, Vol. 52 No. 13, pp. 1715–1720, doi: 10.1016/j.ifacol.2019.11.448.
- Wamba, S.F. and Queiroz, M.M. (2022), "Industry 4.0 and the supply chain digitalisation: a blockchain diffusion perspective", Production Planning & Control, Vol. 33 No. 2–3, pp. 193– 210, doi: 10.1080/09537287.2020.1810756.
- Wamba, S.F., Queiroz, M.M. and Trinchera, L. (2020), "Dynamics between blockchain adoption determinants and supply chain performance: An empirical investigation", International Journal of Production Economics, Vol. 229, p. 107791, doi: 10.1016/j.ijpe.2020.107791.
- Wang, H., Zhang, M., Ying, H. and Zhao, X. (2021), "The impact of blockchain technology on consumer behavior: a multimethod study", Journal of Management Analytics, Vol. 8 No. 3, pp. 371–390, doi: 10.1080/23270012.2021.1958264.
- Wang, Y., Han, J.H. and Beynon-Davies, P. (2019a), "Understanding blockchain technology for future supply chains: a systematic literature review and research agenda", Supply Chain Management: An International Journal, Vol. 24 No. 1, pp. 62–84, doi: 10.1108/SCM-03-2018-0148.
- Wang, Y., Singgih, M., Wang, J. and Rit, M. (2019b), "Making sense of blockchain technology: How will it transform supply chains?", International Journal of Production Economics, Vol. 211, pp. 221–236, doi: 10.1016/j.ijpe.2019.02.002.
- Wang, O. and Scrimgeour, F. (2023), "Consumer adoption of blockchain food traceability: effects of innovation-adoption characteristics, expertise in food traceability and blockchain technology, and segmentation", British Food Journal, Vol. 125 No. 7, pp. 2493–2513, doi: 10.1108/BFJ-06-2022-0466.
- Warshaw, P. R., & Davis, F. D. (1985), "Disentangling intention and expectation", Journal of Experimental Social Psychology, Vol. 21, No. 3, pp. 213-228.
- Wong, L.-W., Leong, L.-Y., Hew, J.-J., Tan, G.W.-H. and Ooi, K.-B. (2020a), "Time to seize the digital evolution: Adoption of blockchain in operations and supply chain management among Malaysian SMEs", International Journal of Information Management, Vol. 52, p. 101997, doi: 10.1016/j.ijinfomgt.2019.08.005.
- Wong, L.-W., Tan, G.W.-H., Lee, V.-H., Ooi, K.-B. and Sohal, A. (2020b), "Unearthing the determinants of Blockchain adoption in supply chain management", International Journal of Production Research, Vol. 58 No. 7, pp. 2100–2123, doi: 10.1080/00207543.2020.1730463.
- Xu, J., Guo, S., Xie, D. and Yan, Y. (2020), "Blockchain: A new safeguard for agri-foods", Artificial Intelligence in Agriculture, Vol. 4, pp. 153–161, doi: 10.1016/j.aiia.2020.08.002.

- Yadav, V.S., Singh, A.R., Raut, R.D. and Govindarajan, U.H. (2020), "Blockchain technology adoption barriers in the Indian agricultural supply chain: an integrated approach", Resources, Conservation and Recycling, Vol. 161, p. 104877, doi: 10.1016/j.resconrec.2020.104877.
- Yen, D. C., Wu, C.-S., Cheng, F.-F., & Huang, Y.-W. (2010), "Determinants of users' intention to adopt wireless technology: An empirical study by integrating TTF with TAM", *Computers in Human Behavior*, Vol. 26, No. 5, pp. 906–915. <u>https://doi.org/10.1016/j.chb.2010.02.005</u>
- Yiannas, F. (2018), "A New Era of Food Transparency Powered by Blockchain", Innovations: Technology, Governance, Globalization, Vol. 12 No. 1–2, pp. 46–56, doi: 10.1162/inov a 00266.
- Zeng, M. and Lu, J. (2021), "The impact of information technology capabilities on agri-food supply chain performance: the mediating effects of interorganizational relationships", Journal of Enterprise Information Management, Vol. 34 No. 6, pp. 1699–1721, doi: 10.1108/JEIM-08-2019-0237.
- Zhang, X., Sun, P., Xu, J., Wang, X., Yu, J., Zhao, Z. and Dong, Y. (2020), "Blockchain-Based Safety Management System for the Grain Supply Chain", IEEE Access, Vol. 8, pp. 36398–36410, doi: 10.1109/ACCESS.2020.2975415.
- Zhao, Guoqing; Liu, Shaofeng; Lopez, Carmen; Lu, Haiyan; Elgueta, Sebastian; Chen, Huilan; Boshkoska, Biljana Mileva (2019). Blockchain technology in agri-food value chain management: A synthesis of applications, challenges and future research directions. Computers in Industry, Vol. 109, pp. 83–99.

Annexure



Figure A1 – Path analysis diagram of the conceptual framework (Authors own creation)

The Intention of Adopting Blockchain Technology in Agri-Food Supply Chains: Evidence from an Indian economy

Abstract

**Purpose** -This research explores the underlying intention behind using blockchain technology (BLCT) in the agri-food supply chain (AFSC). This is achieved by employing a conceptual framework based on Technology Acceptance Models (TAM) that considers various factors influencing user behavior towards implementing this technology in their practices.

**Methodology-** The conceptual framework developed is empirically validated using Structural Equation Modeling (SEM). A total of 258 respondents from Agri -food domain in India were involved in this survey, and their responses were analyzed through SEM to validate our conceptual framework.

**Findings-** The findings state that food safety and security, traceability, *transparency*, and *cost* highly influence the intention to use BLCT. Decision makers of the AFSCs are more inclined to embrace BLCT if they perceive the usefulness of the technology as valuable and believe it will enhance their productivity.

**Implications-** The study contributes to the existing literature by providing thorough examination of the variables that influence the intention to adopt BLCT within the AFSC. The insights aim to benefit industry decision-makers, supply chain practitioners, and policymakers in their decision-making processes regarding BLCT adoption in the AFSC.

**Originality/Value-** The current study investigates how decision-makers' perceptions of BLCT influence their intention to use it in AFSCs, as well as the impact of the different underlying factors deemed valuable in the adoption process of this technology.

**Keywords:** Blockchain (BLCT); Food security and safety; Agri-food supply chain (AFSC); Technology Acceptance Models (TAM); structural equation modelling (SEM)

## 4. Introduction

Food and agriculture are crucial for the Indian economy, but traditional technologies remain prevalent. With an estimated population of 9.6 billion by 2050, the food supply must increase by 70% (FAO 2009; Gardas et al.,2017). Blockchain technology (BLCT) is a potential solution to maintain traceability and transparency in the agri-food supply chain. Modern companies require transparency in food product history and secure information throughout the supply chain (Dabbene et al. 2014). Globalization and increased competition have strained the agri-food supply chain, leading to increased competition and a more complex supply chain. BLCT can help provide proof and authenticity of sources.

BLCT can initiate many changes in the industry (Agrawal et al., 2023). It can change the business models of existing firms, change the level and nature of demand, affect the competitive positions of

various players in the industry, and have a profound impact on the market (Lanzolla et al., 2012; Saurabh et al., 2023). Technology has proven effective as information moves through the SC (Guggenberger et al.,2020; Ali et al., 2021). As a new decentralized and distributed technology, BLCT can transform a wide range of industries by enabling transactions to be recorded and verified in a secure, transparent, and immutable manner. It can address major supply chain management (SCM) issues like visibility, traceability, and transparency (Ahmed et al., 2022). The concept of Blockchain for Good (B4G) gives rise to novel applications that can drive the United Nations Sustainable Development Goals (Trollman et al., 2022).

Integrating BLCT in AFSC can provide innovative solutions for product traceability and food safety (Feng et al., 2020; Kumar et al., 2023a). As a result, many of the current challenges of food safety, security, and transparency faced by SCs can be resolved (Gardas et al., 2019). The immutable and decentralized nature of the BLCT helps to eliminate information asymmetry in SCs by storing data in the form of blocks, enabling the storage and trading of tangible and intangible assets with increased efficiency, reducing risks, and lowering costs (Li. et al., 2021). By providing direct linkages between farmers, retailers, customers, and producers, BLCT helps to restructure AFSCs, improving food quality and transparency and minimizing food fraud and wastage (Li et al., 2021; Collart and Canales, 2022). According to Wang et al. (2021), a BLCT tracing system increases SC transparency and process management and reduces intermediary costs, improving SC service and confidence. With the importance of Blockchain technology in advancing AFSC to greater transparency and traceability, it still faces issues in adoption; thus, the research question still needs to be addressed: *What factors are perceived as helpful in adopting the BLCT in AFSC*?

Successful adoption of BLCT in AFSC will confer a competitive edge and unparallel opportunities to the sector. A significant number of practitioners and scholars point out the BLCT's ability to transform SCs (Schmidt and Wagner, 2019; Wang et al., 2019b; Chang et al., 2020; Wamba et al., 2020) and (Zhang et al., 2020; Ali et al., 2021; Mangla et al., 2021; Vikaliana et al., 2021) studied BLCT in many agri-food industries to understand the implementation benefits and challenges. However, the scarcity of information and an inability to apply BLCT to widely spread applications have hindered such applications, specifically in developed economies. Also, gaining acceptance from key stakeholders and decision-makers remains a pivotal factor in successfully implementing such innovations. Therefore, it is critical to establish the intention to use BLCT in AFSC to understand the factors influencing its adoption and develop strategies for promoting its successful implementation.

And thus, this study intends to address the second research question: *How do the factors identified influence the intention to adopt BLCT in AFSC?* 

Integrating BLCT into the AFSC requires user acceptance and understanding of the factors they perceive as essential for adoption. To ensure the benefits of BLCT, AFSC decision-makers must ensure it benefits their organization and secures consumer trust. To achieve this, a comprehensive assessment of user perceptions and attitudes towards the new technology is necessary (Saha et al., 2023; 2023a). Over time, numerous studies have explored this topic with varying degrees of detail, including seminal works authored by (Venkatesh et al., 2003; 2012; Queiroz and Wamba, 2019a). The Technology Acceptance Model (TAM) is a theoretical framework used to evaluate users' willingness to accept and utilize new technology. The original TAM model, proposed by Davis in 1989, has evolved into different versions, each with unique features. It focuses on perceived usefulness and ease of use, making it a straightforward and practical model for researchers and practitioners studying and improving user acceptance of technology.

User acceptance of the BLCT and its influencing factors, play an important role in the technology's adoption. By considering different technology adoption models and independent variables that influence user acceptance of BLCT, it is possible to predict the level of user acceptance of BLCT in the AFSC domain. Furthermore, it is crucial to understand the decision makers' intention in adopting BLCT, considering this technology's disruptive potential (Wamba and Queiroz 2020) and its unprecedented impact on SCs. Though recent research (Yadav et al., 2020; Liu et al., 2021; Trollman et al., 2022; Kumar et al., 2023a) investigated the potential and factors of BLCT in AFSC, the factors contributing to its widespread acceptance and intention to acknowledge the technology have been overlooked. While few authors (Kamble et al., 2019; Nayal et al., 2021; Paul et al., 2021) emphasized its impact on SC, they did not identify the key factors influencing decision -makers' intentions for BLCT adoption, resulting in gaps in understanding the SC dynamics and stakeholders' perspective. Thus, this study investigates the factors influencing decision-makers intentions to use BLCT in AFSC. The study conducts empirical research using SEM and proposes a conceptual framework based on TAM to study the intention to use BLCT and shed light on the following research objectives (ROs):

*RO1:* To explore the factors helpful in adopting the BLCT in AFSC. *RO2:* To examine the user intention to adopt BLCT in AFSC.

Since blockchain is still in the inception stage and adoption of the technology remains a challenge, this research delves deeper into identifying and understanding the factors that influence the adoption

and intention to use BLCT in AFSC. To achieve this, the study is based on TAM, which prioritizes perceived usefulness (PU) and perceived ease of use (PEU) as predictors of intentions to use (ITU). TAM is used to build the study's conceptual framework, which identifies critical factors impacting ITU while also studying their effects on the TAM construct to gain better insights toward successful implementation. These findings will enable the decision-makers to develop more robust strategies to integrate BLCT across the AFSC effectively. By understanding the factors that drive technology adoption, tailored implementation plans can be fostered for the Indian AFSC.

The research paper is laid out in the following pattern. Section 2 discusses the overview of the relevant literature. Section 3 presents the conceptual framework's development and the formulation of the hypotheses. Section 4 addresses the research design. In Section 5, the data analysis and findings are presented. Section 6 includes a discussion and theoretical and practical contribution of the study. Finally, section 7 concludes the paper with a discussion on future directions.

#### 5. Literature Review

#### 5.1. Blockchain technology and AFSC

Integrating BLCT in AFSC offers a comprehensive solution to trace, track and monitor the entire length of the supply chain (Kayikci et al., 2022; Srivastava et al., 2022), ultimately leading to greater transparency and trust (Liu et al., 2022; Sharma, 2023). While traditional SC have long struggled with food safety concerns, BLCT can address these issues by increasing visibility throughout the chain (Kshetri 2019). Research studies conducted on BLCT and SC integration revealed that adopting this technology improves transparency, authenticity, and real-time transactions (Mukherjee et al., 2021; Feng et al., 2020; Xu et al., 2020). Furthermore, (Sunny et al., 2020) suggest that transparency and traceability are critical factors that impact a logistics organization's overall performance - making it essential for businesses to prioritize integrating BLCT into their SCs. Providing complete visibility upstream and downstream of AFSC-BLCT helps establish accountability across all supply chain stages while enhancing its trustworthiness (Rogerson and Parry, 2020). More and more value chains are incorporating BLCT to make them flexible and strengthen customer ties (Murki et al., 2018). This is a major movement in the rapidly expanding food industry, where BLCT is seen as the best option for tracking and changing. The potential uses of BLCT in the agri-food value chain have been the subject of several studies (Zhao et al., 2019). One such study used RFID and blockchain to create a system to track the agri-food supply chain in China (Tian. 2016).

Further applications of blockchain technology include an IoT software connector, a distributed ledger manufacturing supply chain that is blockchain-ready, and a supply chain traceability solution that is both transparent and decentralized. By facilitating the transfer of more precise and trustworthy data between producers and consumers, BLCT has shown to be more effective than conventional approaches in food supply chains (Zhao et al., 2019). Because of this, the food sector is now more efficient and cost effective. Ultimately, the food sector is realizing the importance of BLCT, which provides a safer and more effective method to monitor and record food supply chains. Businesses may improve their operations and cut expenses by using BLCT in several industries, including supply chains, logistics, and agriculture.

As the research shows, integrating BLCT in AFSC is an innovative move that promises many advantages. With its ability to improve transparency and trust throughout the SC, BLCT has been proven beneficial for AFSC. However, it is worth noting that despite this promising development, the intention to use BLCT in AFSC is still relatively new and requires further exploration before its potential benefits can be fully realized. For decision-makers to effectively integrate BLCT into their AFSC, they must first believe in its perceived usefulness and ease of use of the technology. This requires a deeper understanding of how BLCT works, and which factors benefit and fit within their specific industry context.

#### 5.2. Research Gaps

BLCT adoption and intention to use in SC has been studied by various authors such as Wamba & Queiroz (2020) in the Brazilian SC, Queiroz et al. (2021) in logistics and SC, Karamchandani et al. (2022) for the service SC, Sternberg et al. (2021) in inter-organizational SC. While several models on BLCT adoption have emerged in recent years within the agriculture and food sectors context, studies such as (Saurabh and Dey, 2021; Ronaghi, 2021; Paul et al., 2021; Liu et al., 2021; Yadav et al., 2020; Susanty et al., 2021), did not fully explore critical issues that drive its widespread acceptance. Queiroz et al. (2021) conducted a comprehensive study on adopting BLCT, predicting the likelihood of implementing BLCT, while Queiroz and Wamba (2019) emphasized that performance expectancy is one of the enabling conditions of BLCT adoption. Interestingly enough, both US-based firms and Indian organization were found to be influenced by this factor. Furthermore, Wong et al. (2020a) focused on additional considerations related to adopting BLCT - specifically costs incurred during the implementation phase, regulatory support from authorities, top management involvement & encouragement towards its utilization within organization frameworks. The results indicated that cost implications had widespread influence over intentions regarding adoption rates

among respondents surveyed. Several scholarly studies have delved into the multifaceted ways in which BLCT can impact and influence the performance of SCs, as observed by Nayal et al. (2021), Masudin et al. (2021), and Paul et al. (2021). Vasan et al. (2023) has explored the farmers' belief in adopting technology, but the entire supply chain has not been discussed. These investigations shed light on a complex interplay between BLCT and SC but fail to uncover the potential factors that are perceived useful by decision makers and lead to the intention to use the technology. Also, the intention to use BLCT in Indian AFSC has not been explored much through TAM.

Although few studies have been conducted on the potential for blockchain adoption in supply chain management, a more comprehensive analysis of users' intentions to use BLCT is necessary. Understanding the factors that motivate users to embrace this technology will be crucial in driving its widespread adoption throughout AFSC. To truly integrate blockchain as a solution, it is essential to examine BLCT's perceived usefulness and user motivation. To shed light on these key aspects of blockchain implementation, empirical research has focused on ITU BLCT in AFSC and variables deemed useful by prior literature. Notably, cost (COS), food safety and security (FSS), transparency and traceability (TT) have emerged as salient concerns related to adopting this technology within supply chains. A TAM framework has been adopted to comprehend how independent variables affect both PU and PEU of the technology, ultimately leading towards higher rates of ITU-BLCT in AFSC.

#### 6. Hypotheses Development and Conceptual Framework

#### 3.1 Technology Acceptance Models (TAM)

Technological advancements continue to be significant in business, and acceptance of technology is required to implement them successfully. In light of this reality, companies must deeply understand the factors contributing to successful implementation. To elucidate these complexities, this study relies on TAM, originally proposed by Davis (1985), as an instrumental tool for assessing technological adoption. Extensive research has been conducted exploring different aspects of TAM over the years, including but not limited to the works by (Davis, 1985, 1989; Venkatesh and Davis, 2000; Venkatesh et al., 2003; Venkatesh and Bala, 2008; Venkatesh, Thong and Xu, 2012). By examining individual or organizational reasons behind technology uptake through applying the TAM framework, we can gain insight into how best to facilitate its utilization within businesses.

TAM has long been a cornerstone in the field of technology acceptance research. Although numerous variations, such as TAM2 and extended TAM models, have emerged, the original model still holds significant weight due to its extensive validation. Venkatesh and Davis (2000) argue that while newer versions introduce new variables like social influence and cognitive instrumental processes, their complexity may limit practical application. This debate was put to the test by Wu and Wang's (2005) study comparing the usefulness of both original TAM versus extended version- results showed that although expanded versions explained more variance in data, the simplicity provided by the original model made it an effective tool for predicting technology acceptance patterns with reasonable accuracy. Ultimately, this affirms support for utilizing original TAM as an undeniably successful methodology for seeking insights into understanding how users accept new technologies effectively under varying contexts.

TAM explores the link between PU, PEU, and ITU characteristics. Using TAM, Davis et al. (1989) proposed that PU, and the attitude toward use, influence technology adoption. PU assesses an individual's attitude towards technology. This attitude usually reflects how much benefit they expect from using the technology and how much effort they think they will have to expend to use it. Furthermore, it is anticipated that PEU has a direct effect on PU. PEU represents how easy it is for an individual to use the product/service. A product with a high PEU score will be perceived as easy to use and therefore more likely to be adopted by users. ITU is the willingness of an individual to use a technology once the individual is aware of it. It is determined by measuring the level of motivation to use the technology in question as well as a person's attitude toward using such technology.

# 4.2 Conceptual framework

The conceptual framework has been adapted from Venkatesh et al. (2003). The framework based on the TAM concept studies the perceived usefulness (PU) of the BLCT in the AFSC. The framework consists of three variables: COS, FSS, and TT, identified from the literature review as influential on the adoption and ITU the BLCT technology. Figure 1 shows the proposed conceptual framework. While the model is comprehensive, it may not cover all factors influencing the adoption of BLCT. However, to ensure parsimonious measurement instrumentation and considering the survey length, the study was limited to only three variables FSS, COS, and TT.



"\*COS- Cost; FSS- Food safety and security; TT- Transparency and traceability; PEU- Perceived ease of use; PU- Perceived usefulness; ITU-Intention to use"



#### 3.1 Cost

The cost of adopting a new technology goes beyond just the financial aspect. It also includes emotional effort, time, and energy investment to learn and integrate it into existing systems. However, this initial expense can eventually save money by streamlining operations and reducing transactional costs in the long run. Decision-makers are more likely to invest in technologies that will significantly reduce operational expenses over time. One such innovative solution proposed by Li et al. (2021) is using BLCT for automation, which eliminates manual involvement, intermediaries, and paperwork, thus minimizing costs associated with supply chain management. Incorporating smart contracts further helps optimize procurement processes as suppliers receive prompt payment upon delivery confirmation while reducing human interaction between parties. Vu et al., (2021) demonstrate that such automated procedures streamline workflows and cut down on transaction expenditures like distribution expenses or data encryption fees, thereby boosting overall efficiency levels. A recent study conducted by Du et al. (2020) found that adopting blockchain technology can lead to improved efficiency and cost reduction. Building on this, Ko et al. (2018) discovered that the costs associated with implementing new technologies directly impact users' perception of usefulness, ultimately influencing their decision to adopt or reject these innovations. However, the perceived cost might not directly influence users' beliefs about the usefulness of the technology or their intention to use it. These aspects are more related to the perceived benefits and value that the technology provides. The influence of cost is mediated through the perceived ease of use of the technology rather than directly impacting perceived usefulness or intention (Luarn et al., 2005; Amoako-Gyampah, 2007; Yen et al., 2010). So, we propose the hypothesis:

H1. Cost positively influences the PEU of BLCT.

# 3.2 Food Safety and Security

Food safety and quality assurance are more challenging with the increase in the global flow of goods (Fischer, 2013). Food safety and security methods include handling, processing, preparing, and storing meals in a way that reduces the likelihood of people becoming ill due to contaminated foods. Thus, FSS refers to delivering a product that has not been contaminated, damaged, or tampered (Marucheck et al., 2011). BLCT helps make the FSC more visible, traceable, and accountable, ensuring safety and security (Gupta and Shankar, 2023). BLCT minimizes food fraud by ensuring that records are transparent and traceable (Danese et al., 2021). As financial records for all operations are documented and stored in the chain, it can also speed up auditing and conflict resolution (Chang et al., 2020). In addition, the BLCT allows parties to update product information in near real-time,

minimizing the possibility of cross-contamination and improving responsiveness. It also assists in providing food safety status in real-time to all SC decision-makers, thereby providing a secure, distributed, transparent, and collaborative information system. In their research, Roy et al. (2020) state that food safety and security are essential in earning SC decision-maker trust. Trust leads to the PEU of technology, thereby increasing the PU and ITU the technology leading to the following hypotheses:

H2. Food safety and security positively influence the PEU of BLCT.

- H3. Food safety and security positively influence the PU of BLCT.
- H4. Food safety and security positively influence the ITU BLCT for AFSC.

# 3.3 Traceability and Transparency

SC traceability and transparency has become increasingly important to sustainable management (Mollenkopf et al.,2022) A traceable and transparent SC demonstrates that an organization is honest and upfront about how it conducts business (Kafetzopoulos et al., 2023) A blockchain-based traceable SC increases visibility, improves quality controls, and reduces risk (Chavalala et al., 2022). Transparency builds trust among suppliers, companies, and customers. Astill et al. (2019) state that traceable and transparent food production systems must facilitate data exchanges between stakeholders as this will lead to expediting processes and a reduction of transaction times. Kamath (2018) discusses how BLCT helps trace and identify the SC's origin and path and build SC decision-maker trust. Being an immutable technology, BLCT reduces human intervention and is thereby helpful in food recall and food wastage (Duan et al., 2020). The traceability and transparency benefit of BLCT leads to the PU and ITU of the technology. Hence, we suggest the following hypotheses:

H5. Traceability and transparency positively influence the PEU of BLCT.

H6. Traceability and transparency positively influence the PU of BLCT.

H7. Traceability and transparency positively influence the ITU BLCT for AFSC.

## 3.4 Intention to Use

Perceived adoption intention refers to a user's active intention to adopt a specific technology (Queiroz & Wamba, 2019). Among the main strands of the technology management literature, an important focus is on adoption intention, which explains how individuals or decision-makers respond to a given technology that may lead to its actual usage. Research has shown that adoption intention is one of the most significant predictors of technology usage (Venkatesh et al., 2012) and competitiveness (Kshetri, 2018). In their research, Davis et al. (1989) define perceived usefulness as the degree to which a user believes employing a given framework would improve productivity and

efficiency. The literature shows that perceived usefulness significantly influences users' attitudes and intentions. Currently, BLCT is being used in a wide range of sectors. The users are more likely to be positive towards BLCT if they perceive it as useful and can increase efficiency. "Perceived ease of use" refers to how easily someone believes a particular technology or system is used (Davis, 1989). The literature shows that PEU positively affects PU (Venkatesh and Davis, 1996). Increasing user perceptions of BLCT as useful will make users more positive (Liu et al., 2021). Maintaining competitiveness, easy tracking and tracing of products, monitoring fair trade, cooperating with multiple participants, and building SC decision-maker trust may influence the use of BLCT for achieving benefits for firms and lead to the ITU of the technology. Hence, the followings hypotheses have been formed:

H8. PEU positively influences the PU of BLCT.H9. PU positively influences the ITU BLCT for AFSC.

# 5 Research Design

## 4.1 Questionnaire Design

The study uses in-depth interviews and questionnaires to examine the impact of different factors on the intention to use BLCT. The survey method is used to elicit information about respondents' backgrounds and attitudes as well as their intentions for using BLCT. A five-point Likert scale ("strongly agree to strongly disagree") was selected to measure the items and variables. The survey questionnaire was designed based on the previous literature and were validated by experts (industry and academician). Six variables and 24 items were used in the survey. The constructs utilized in the questionnaire are listed in their sources in Table 1. Six industry and 4 academic experts working in the agri-domain were asked to evaluate the questionnaire and check each construct for its reliability and validity. They reviewed the items for ambiguity, consistency, and relevance to the survey (Kumar et al.,2022). In addition to ensuring the consistency and reliability of the questionnaires for large-sample surveys, the validity and reliability of each scale are also checked.

Variable		Item	Description	References		
Cast	COSI	A dontion aget	"The elements indicate the entire	"Tapscott et al.,		
Cost	COSI	Adoption cost	implementation of BLCT. It	Fischer, (2019);		
		Tashnalagy	helps in reducing the overall	Ivanov et al.,		
	COS2	recimology	AFSC expenses by minimizing	(2019); Yadav and		
		COST	the middle entities. It also helps	Singh (2020); Wong		

Table 1: Variables and Items Table

	COS3	Maintenance	in reducing the operating cost,	et al., (2020a);
	_	cost	manpower cost, and	Sternberg et al., $(2021)$ : Vu et al.
			less in consumable items	(2021); vu et al., $(2021)$ "
			compact data storage sharing	(2021).
	~~~	Transaction	database, faster payments, less	
	COS4	cost	human interaction,	
			rapid decision-making, strong	
			management, and an efficient SC."	
	FSS1	Improve food	"Food Safety and Security	
	1351	quality	suggest that BLCT is an	
	FSS2	Reduce cross	effective technology to avoid	
		contamination	food fraud, reduce cross	"C 1 1
	FSS3	Maintain	contamination, and increase	"Galvez et al., $(2018)$ . Tore et al.
Food Safety		records	to retrieve the required	(2010), Tall et al., $(2018)$ : A still et al.
and Security			information confine	(2010); Astin et al., (2019): Crevdt and
			contaminated products from	Fischer. $(2019)$ ; Creyar und
		Ensure security	suppliers, and minimize the	Duan et al., (2020);
	FSS4		products recall range. BLCT	Lin et al., (2020)."
			also provides data security from	
			cyber threat and food fraud. The	
			data are stored in immutable	
			blocks and cannot be tampered."	
	TT1	Reduce Iood	BLC1 helps to manage	
		Reduce food	tracking of products Firms can	
	TT2	fraud	track the information on the	
		Easier food	storage life of food products	
Traceability	113	recall	better thereby improving profits.	
and	TT4	Improved	BLCT can also assist in	"Kamath, (2018);
Transparency	117	tracking	promoting economic	Astill et al., (2019);
			sustainability of the AFSC by	Jen Yin Yeh et al.,
			reducing food waste and food	(2019)."
		Increase in SC	process the food and track the	
	TT5	decision-	inaccessible data which can be	
		maker trust	utilized to improve SC	
			procedures and build SC	
			decision-maker trust."	
	PEL1	ease of use	"BLCT contributes to address	"Venkatesh et al
Perceived			the complex industrial systems	2003. $2012)$ .
	PEU2	ease of	(CISs) challenges. BLCT	Christopher Lee et
ease of use		understanding	reduces the intricacy of cash data	al., (2019); Queiroz
		Commed:11.1	storage and recovery system of	et al., (2019)."
	PEU3	Compatible to	finance and other departments	
		use	Operational usefulness has a	

			positively influence on the BLCT's ease of use."			
	PU1	Improve performance	"BLCT enhances the AFSC's	"Variath (2019).		
Perceived	PU2	Simplify process	performance and simplifies the complex AFSC network by	Christopher Lee et		
usefulness	PU3	Improve safety and security	tracking and tracing the entire SC. Thereby improving safety and security. It also maintains	al., (2019); Creydd and Fischer, (2019); Kshetri, (2019); Hew et al., (2020); Gao et al., (2020); Vu et al. (2021)."		
	PU4	Increase traceability	and stores records in an immutable form which is easy to			
	PU5	Maintain records	access but difficult to tamper."	v u et al., (2021).		
	ITU1	Innovative technology	"BLCT is an innovative distributed technology that helps	"Kamath, (2018); Kamilaris et al.,		
Intention to use	ITU2	SC decision- maker satisfaction	build trust, reliability, and satisfaction among the SC decision-maker as it can provide	(2019); Kshetri, (2019); Lin et al., (2020);		
	ITU3	Better transaction facility	farm to fork data. BLCT helps to build a better and smooth- running transaction facility."	Schinckus, (2020); Wang et al., (2020)."s		

# 4.2 Sampling and data collection

Data are collected from the questionnaires provided to Indian agri-food practitioners working in the domain for at least 2.5 years and having a basic knowledge of BLCT. Data are collected between June and August 2022 from multiple agro-based Indian companies to generalize the findings to a broader range of companies (Nayal et al., 2021). The purposive sampling technique is used as the sampling method to identify the interviewee. Respondents are also addressed individually whenever possible to enhance the survey response rate. Five hundred professionals from the agribased industry, including vegetables and fresh fruits, dairy, and beverage industry, knowing the technology implementation benefit, are invited to participate in the study. Of 550 professionals, 258 provided useful answers (a response rate of 46.9%). The average participant's ages are between 36 and 55 years old. It has been observed that the male respondents outweigh the female respondents, with the majority of participants aged 25 to 75 years. The participants' demographic characteristics are categorized in "Table 2".

			()
Items		Ν	Percentage
Age	25-35	96	37.2
	36-55	104	40.3
	56-75	58	22.5
	Total	258	
Gender	Male	140	54.26

**Table 2**: "Demographic profile of the respondents" (N -258)

	Female	118	45.73
	Total	258	
Educational	UG	98	38
Qualification	PG	107	41.5
	PhD	53	20.5
	Total	258	
Years of Experience	0-5	87	33.7
	5-10	58	22.5
	10-15	62	24
	15-20	51	19.8
	Total	258	

### 4.3 Non response and common method bias (CMB)

Non-response bias was mitigated by employing exploratory and subjective approaches (Armstrong and Overton, 1977). To evaluate the difference in means for every scale, the t-test is conducted for the early and late responses of the participants for a non-response bias test (Karamchandani et al., 2020). No significant difference in the scale items for each construct was found for this study. To prevent the common method bias (CMB), the questionnaire was conscientiously designed and tested with expert practitioners. Anonymity was ensured during data collection, and respondents were assured that there were no right and wrong answers, emphasizing the significance of truthful answers (Podsakoff et al., 2003). It is important to check and control CMB when the data from all the constructs in a model are collected from a single respondent (Podsakoff and Organ, 1986). To minimize the effect of common method variance, the sequences of the survey questions for every respondent were randomly assigned, and the respondents were asked to give honest feedback while they were also assured that their responses would remain anonymous (Podsakoff et al., 2003). In addition, Harman's single-factor analysis was performed to evaluate and mitigate the potential presence of biases. The results showed a single factor of less than 40% of the variance, suggesting no CMB. Even though common method bias affects the linear relationships, it seems to have fewer effects on the analysis of interactions between variables.

#### 6. Data Analysis and findings

SEM is the statistical method used for hypotheses testing. With SEM, one can validate the suitability of TAM in analyzing BLCT's ITU. SEM helps specify, estimate, and evaluate models of linear relationships among variables in relation to a relatively small number of unobserved variables (Shah et al.,2006). The data analysis was carried out using AMOS software, which has all the tools necessary for creating and examining SEM path diagrams (Nayal et al., 2021) and analyzing SPSS files (Mangla et al., 2020). IBM-SPSS Amos is robust software for SEM that allows users to validate

their research and hypotheses by expanding conventional multivariate analysis techniques, such as regression, factor analysis, correlation, and analysis of variance. Using SPSS-Amos, one can develop intricate attitudinal and behavioral models that depict complex relationships more precisely than traditional multivariate statistical methods through an intuitive graphical or programmatic user interface (IBM, 2017; Dash et al., 2021).

SEM relies on path diagrams to provide a clear picture of the relationships between variables (Ullman and Bentler, 2012). For the SEM model fit, 100 sample size is qualified as fair; the current sample size is 258 beyond the threshold value (Ganbold et al., 2021). "Exploratory factor analysis (EFA)" and "confirmatory factor analysis (CFA)" were performed before the SEM analysis. EFA measures the correlation between constructs in a dataset by analyzing the correlations between them. It also reduces the broad data set to smaller set. Bartlett's test of sphericity and "Kaiser-Meyer-Olkin (KMO)" helps to understand the suitability of the data. The results show that the KMO value is 0.851(value>0.7), which is satisfactory. The measurement model is validated using CFA to identify how well the conceptual framework reflects the data. The results show that the Chi-square test result is 1.716 (< 3.0), the "Goodness fit index (GFI)" is 9.51 (>9.50), the "comparative fit index (CFI)" is 0.951, and the "Root mean square error of approximation (RMSEA)" is 0.053, which suggests that the model is a close fit.

# 5.1. Structural model assessment

To evaluate how reliable or consistent the questionnaires are, they are tested for dependability. The analysis reveals the extent to which measurement inaccuracies impact (or do not affect) the gathered data. In general, it is acknowledged that data can be regarded as credible if Cronbach's alpha is greater than 0.70 (Teo et al., 2015; Kumar et al. 2022). Table 3 shows that all the item loadings exceeded the 0.70 threshold, providing evidence of convergent validity (Jiang et al., 2020). Therefore, the items for each construct and the complete questionnaire have good internal consistency. Composite reliability (CR) in Table 4 shows that all six constructs are higher than 0.7, indicating that all the six are valid. Tabachnica and Fidell (2007) state that if the average variance extracted (AVE) is larger than 40%, the measurement questions represent the features of each study variable in the model. Each variable in Table 3 has an AVE value higher than 0.5, denoting convergent validity of the scale. Convergent validity can be further examined by analyzing the AVE t-values for factor loadings, and composite reliability (construct reliability) (Chau, 1997; Fornell & Larcker, 1981).

Discriminant validity can be evaluated by examining the square root of AVEs for each latent variable with its correlation to the other constructs (Fornell and Larcker, 1981). Table 4 shows the correlations between the constructs and demonstrates the discriminant validity of the variables. The

correlation coefficient is higher than the square root of the AVE, and the AVE value for every construct is more than 0.5, suggesting that the constructs have significant discriminant coefficients, and that the model's intrinsic quality is ideal.

The path diagram of SEM prepared in "AMOS-20.0" is shown in Figure 2. The results of the SEM and the path analysis of the conceptual framework are displayed in Table 5. They include "the path, coefficient, standard deviation, t-value, and p-value". The findings show that all nine hypotheses support, which leads to the conclusion that all the independent variables, COS, FSS, and TT, positively influence the dependent PEU, PU, and ITU. Statistical significance was found across all paths in the framework, with standardized "path coefficients ranging from 0.1 to 0.4". The path coefficient( $\beta$ ), (+/-1) indicates the degree of change in the outcome variable for each one unit change in the predictor variable. P values (<0.05) determine that the hypothesis results are statistically significant. The value shows a positive relationship between the independent variables (COS, FSS, and TT) and the dependent variables (PEU, PU, ITU). The findings reveal that food safety and security is perceived as the most important useful factor influencing the ITU the technology, followed by cost and traceability.



**Figure 2:** Measurement model (\*\*\* p<0.001, \*\* p<=0.01, \*p<= 0.05)

Constructs	Items	Loading factors	α	CR	AVE
	PU3	0.961			
Perceived	PU4	0.943			
usefulness	PU2	0.839	0.946	0.946	0.596
	PU5	0.811			
	PU1	0.804			
D	PEU3	0.861			
Perceived ease	PEU1	0.842	0.007	0 000	0.602
of use	PEU2	0.793	0.88/	0.888	0.602
	PEU4	0.757			
	TT1	0.952			
Transparency	TT5	0.923		0.9	0.539
and traceability	TT2	0.730	0.899		
	TT3	0.678			
	TT4	0.638			
	FSS3	0.994		0.913	0.543
	FSS2	0.935	1		
Food safety and	FSS5	0.719	0.911		
security	FSS1	0.674			
	FSS4	0.673			
	COS3	0.798			
Cost	COS2	0.729	0.751	0.756	0.523
	COS4	0.618			
	ITU4	0.780			
	ITU3	0.778		0.834	
intention to use	ITU2	0.754	0.819		0.621
	ITU1	0.614	1		
	ITU5	0.584	1		

**Table 3:** Items; loading factors; Cronbach's alpha (*α*); Composite reliability (CR); average variance extracted (AVE)

# Table 4: Discriminant validity

Factor	PU	FSS	ТТ	ITU	PEU	COS
PU	0.544					
FSS	0.53	0.776				
TT	0.393	0.348	0.734			
ITU	0.44	0.51	0.42	0.737		
PEU	0.343	0.311	0.381	0.327	0.723	
COS	0.14	0.221	0.214	0.173	0.297	0.788

 Table 5: Path coefficients, standard error, t-statistics, and p-values

"Hypot heses"	"Path"	"Coefficient"	"Standard error"	"t- statistics"	"P- values"	"Supported/ Not supported"
H1	COSPEU	0.284	0.1	2.837	**	Supported
H2	FSSPEU	0.2	0.081	2.471	*	Supported
H5	TTPEU	0.245	0.062	3.921	***	Supported
H3	FSSPU	0.409	0.063	6.437	***	Supported
H6	TTPU	0.15	0.046	3.269	***	Supported
H8	PEUPU	0.104	0.051	2.027	*	Supported
H9	PUITU	0.132	0.059	2.229	*	Supported
H4	FSSITU	0.248	0.059	4.192	***	Supported
H7	TT-ITU	0.126	0.039	3.204	***	Supported

# Significant at P-value p < 0.05; \*\*\* p < =0.001; \*\* p < =0.01, \*p < =0.05# *t*-statistics > 1;  $\beta$  strongest relationship (0 to 1)

## 5.2. Mediation effect

Mediation analysis in SEM is a statistical method to investigate the underlying mechanisms or pathways through which an independent variable affects a dependent variable. Mediation analysis helps identify one or more intermediate between the predictor and outcome variables and explain their relationship. The results of the mediation effect denote that all the paths (TT-PU-ITU, FSS-PU-ITU, TT-PEU-PU, FSS-PEU-PU) have a partial mediation effect. Partially mediated relationships assume that the mediating variable contributes to some of the relationships between the independent and dependent variables. Partially mediated relationships are not merely characterized by a substantial association between the mediator and the dependent variable but also by direct interactions between them. The results of the mediation effect are shown in Table 6.

Path	Direct effect	Indirect effect	Decision				
TT—PU—ITU	0.245 (0.001)	0.041 (0.009)	PM				
FSS—PUITU	0.374 (0.001)	0.079 (0.009)	PM				

0.210 (0.001)

0.450 (0.001)

TT-PEU-PU

FSS-PEU-PU

I able	6:	Results	01	med	lation	effect

\*\*Significant at P-value < 0.1; parenthesis values (P-values); Partial Mediation (PM)

0.037 (0.003)

0.021 (0.008)

PM

PM

# 8. Discussion

The study identifies that the customer's willingness and the variables they perceived to be useful lead to adopting the technology. The study shows that FSS, COS, and TT positively influence ITU technology. Dwivedi et al. (2016) state that cost is essential in the implementation and intention to adopt the technology. The study hypothesis "H1( $\beta$  = 0.284, T= 2.837, p=0.005)" supports the claim that PEU, PU and ITU of the technology are influenced by cost. The path coefficient COS significantly relates to the predetermined variables PEU and PU. The study of (Wong et al., 2020a; Nayal et al.,2021) agrees with our finding that cost saving significantly impacts the perceived usefulness and ITU of the technology. The analysis of the results indicates that the more the decisionmaker believes in the usefulness of the technology and the value for money it provides, the more the intention to adopt the technology increases. The  $\beta$  value of PEU and PU of the variable COS is less than FSS and TT as BLCT is a high-end technology, and implementation cost as an individual technology can be higher even if implementing it as a group will reduce its initial investment cost (Roeck et al. ,2020). It decreases overall expenses and supports more users using its services than a single user. Also, its safety mechanism is more secure as the users are verified under a certain criterion before it can be used. Customer preferences often lead to processing small volumes of transactions as quickly as possible and at a low cost, which can be possible by adopting BLCT. Chen et al. (2020) state that BLCT will help reduce the network's intermediaries, leading to cost reduction. Jiang et al. (2022) state that customers tend to emphasize whether the transactions can be approved quickly and cost-effectively. By investing in blockchain resources, all trading partners will be able to coordinate their activities on one platform, facilitating instant, transparent transactions that are free of delays (Ali et al., 2021). BLCT is believed to boost security while saving time and money by minimizing documentation processes (Karakas et al., 2021). However, the initial capital required to implement BLCT might cause difficulties for the developing AFSC sector and may limit its adoption.

FSS describes a food product's level of assurance that it will not cause illness or injury during its production, serving, or consumption, which helps customers build trust in the product by reducing their concerns about food products. The variable FSS and the supporting hypothesis "H2( $\beta$  = 0.2, T = 2.471, p = 0.013)"; "H3( $\beta$  = 0.409, T= 6.437, p<0.001)"; and "H4( $\beta$  = 0.248, T= 4.192, p<0.001)". state that FSS positively influences the intention and adoption of BLCT. A higher  $\beta$  value of the PEU and PU of FSS confirms that the FSS plays an important parameter that influences the adoption of the BLCT. The FSS, in particular, has not been discussed in any adoption model.

Nevertheless, Wang and Scrimgeour (2022) highlight that decision maker perception of food quality influences the intention to use the technology. The existing literature (Tan et al., 2018; Duan et al., 2020; Lin et al., 2020) provides a similar view. Integrating BLCT in the AFSC enhances decision confidence as it helps the food manufacturers to share information about the origin, batch numbers, and production dates, as well as promote food safety, certification, and organic products

(Galvez et al., 2018), which act as a motivator for PEU and PU. BLCT enhances the ability to identify potential sources of contamination (Alladi et al., 2019), stop the illegal purchase of pesticides (Leng et al., 2018), and track the manufacturing and expiration date, which in turn helps reduce wastage. For instance, Walmart and Carrefour partnered with IBM's Food Trust blockchain network (Vu et al., 2021) to achieve real-time and end-to-end visibility at low costs.

Similarly, JBS integrated BLCT in their livestock farming to eliminate food fraud in their SC (Saha et al., 2022). These initiatives aim to enhance customers' trust and ensure safety and security in the AFSC. The decision-makers believe that a specific technology will help attain the objective, which leads to the ITU technology.

The variable TT of the SC can help increase decision-maker trust. Tracing the products to their origin has immensely helped to gain the trust of the decision-maker. Integrating BLCT in food traceability and transparency is an identification tool that helps track and trace food products quickly and accurately. The results "H5( $\beta$  = 0.245, T= 3.269, p<0.001)", "H6( $\beta$  = 0.15, T= 3.269, p=0.001)", and "H7( $\beta = 0.126$ , T= 3.204, p=0.001)" are in support of the hypotheses and show significant positive reliability to the adoption and intention to use the BLCT. The analysis supports (Queiroz and Wamba 2019; Wang et al., 2021; Wang and Scrimgeour, 2022), who state that TT is an essential determinant for the adoption of BLCT. TT brings trust and reliability to the system, which supports our analysis and is verified by (Queiroz et al., 2020) but in disagreement with (Queiroz et al., 2019). BLCT provides a reliable source of information about the origin of food products, allowing businesses to differentiate themselves from competitors (Vu. et al., 2021) and is perceived as valuable by decision-makers. However, the PU of TT falls short of FSS, maybe due to the SC members' lack of trust or the availability of other integrated technology like IoT, which is perceived as easier to use. Schulze-Ehlers et al. (2014) state that supplier, processor, and buyer asymmetry leads to a poor supplier-buyer relationship and an unoptimized SC. Li et al. (2022) and Wamba and Queiroz (2022) state that BLCT is ideal for increasing SC traceability, efficiencies, responsiveness, and managing supplier relationships. Adopting BLCT can help companies manage food safety crises more effectively, minimize the impact of food recalls, and enhance their SC productivity. BLCT's ability to track and trace also helps in storing and transporting the food, easing the process. In addition, it can also assist in speeding up the auditing and conflict resolution process, as financial records of all transactions can be documented and stored in the chain (Chang et al., 2020). Thus, the benefits of traceability and transparency are highly perceived to be useful and influence the adoption of BLCT in companies. In addition, the decision-maker will gain confidence in the provider and stop looking

for alternative solutions, which will benefit the company (Villena et al., 2019). The technology's benefits contribute to perceived usefulness, significantly affecting the intention to use BLCT.

The last two Hypotheses (H8) and (H9) analysis show how PU and PEU of the technology positively influence the intention to use the technology. "H8( $\beta$  = 0.104, T= 2.027, p=0.043)" and "H9  $(\beta = 0.132, T = 2.229, p = 0.026)$ " support the hypotheses which are aligned with the findings in Lie et al. (2021) and Ullah et al. (2022). This suggests that the more decision-makers find the usefulness of the BLCT, the more they see the value of the technology and want to adopt it. Furthermore, the H9 has the smallest t value (t=2.027), which implies PEU of technology plays an important parameter in the ITU of the technology. The simpler the technology's ease of use, the better the decision-maker is ready to use it. Comparing our results with Queiroz et al. (2019), shows that trust, transparency, and traceability among the stakeholders and decision-makers influence the intention to adopt BLCT in the developing economy. To fully leverage the power of ease of use for BLCT adoption, organizations that use BLCT should showcase the benefits in their applications to increase customer trust. Governments and private organizations run many pilot projects to implement the technology in the system. In Latin America, the GrainChain startup company used BLCT to track the movement of the grain from farm to market, ensuring that farmers receive fair prices for their crops. Similarly, Agri10X, a startup in India, leverages digital technologies, including BLCT, to provide comprehensive solutions to every phase of the agricultural value chain, aiming to improve farmers' lives. By introducing trust in the system and lowering the cost of online transactions, BLCT also improves the organizations' synergy and efficiencies.

#### 9. Conclusion

This research aims to shed light on the ITU BLCT in AFSC. According to the findings of this study, users intend to use a technology once they perceive its usefulness and ease of use. The study uses an empirical approach based on the proposed conceptual framework on TAM to analyze the ITU and the BLCT in the AFSC domain. The conceptual model was evaluated using SEM, and the results validated the proposed framework. Based on our statistical results, we can conclude that the perceived benefits of BLCT are translated into practical usefulness in three dimensions: food safety and security, transparency and traceability, and cost. According to the study, the perceived value of BLCT among AFSC practitioners in India is not driven by hype but by the knowledge of the perceived benefits. The findings show that practitioners and decision-makers believe in the usefulness of BLCT, which enhances the SC by reducing food losses and wastage and increasing quality and efficiency. The decision-maker believes that integrating BLCT will help to trace and track the product and

identify the origin and flow of the product and any potential sources of contamination occurring in the network. When SC intermediaries can get a good quality product and a visible SC, it helps to build trust between the supplier and consumer, leading to a higher likelihood of technology adoption.

Finally, it is essential to recognize that implementing BLCT cannot be expected to happen overnight, particularly in developing nations where infrastructure and regulatory frameworks are still evolving. However, instead of relying solely on government intervention or aid programs for progress, the global aid community should focus on establishing standards and enabling investors and companies to adopt diverse solutions. This approach will promote platform interoperability and provide potential investors with deeper insights into blockchain-based investment opportunities while emphasizing their societal benefits within the AFSC domain. A well-designed standardization system could enhance transparency across borders by streamlining cross-border transactions and promoting ethical practices among businesses operating in emerging markets- ultimately benefiting both stakeholders involved and facilitating sustainable economic growth globally.

## 9.1. Research Contributions

# 9.1.1. Theoretical Contribution

The current study contributes to both theory and practice. Our research studies the acceptance of blockchain adoption technology by providing a TAM approach practical for innovative technologybased services. We find several factors that positively influence AFSC's intention to use BLCT. Our study highlights the driving factors, FSS, TT, and COS, influencing the intention to use BLCT based on TAM. Our results highlight the importance of including these factors as they help build the lane for adopting the technology.

As theorized by TAM, both PU and PEU significantly influence the intention to use BLCT (Yen et al., 2010). According to the empirical findings of (Kamble et al., 2018), PU is influenced by PEU, and the PU is explained by 28%, suggesting other factors may be at play. Our study shows similar results:  $R^2$  (PEU)= 0.37,  $R^2$  (PU)= 0.21 with three different constructs for PU. The value of  $R^2$  accounted for PU for (COS)= 0.24, (TT)= 0.65, and (FSS)=0.38. The study implies that the higher the perceived benefits of implementing BLCT (saving time, efficiency, and convenience), the more likely users are to use it. The empirical results also indicate that BLCT's characteristics can influence PU and PEU directly; the higher the ease of use and perceived benefit, the higher the chances of intention to use.

Furthermore, these findings highlight the importance of addressing technical considerations and user perceptions when introducing new technology solutions into an organization's workflow. This approach will help foster greater employee acceptance and engagement while maximizing ROI. The contingency perspective in this study advances our understanding of technology acceptance and makes the results relevant to practice.

## 9.1.2. Managerial Contribution

The emergence of BLCT technology has been a significant development in agriculture and food security. However, as this technology is still in its inception stage, it requires a more comprehensive understanding and awareness among manufacturers, policymakers, governments, and individuals for its effective implementation on a wider scale. BLCT demands higher technical knowledge from system designers or developers than traditional agricultural technologies, so they must simplify the setup process by providing clear instructions with an accompanying manual. Additionally, managers should collaborate closely with technology service providers to mitigate the high computing costs of adopting this new system. In particular, initial investment costs may be prohibitively expensive for some companies, which could deter them from investing in such advanced systems; therefore, reducing these expenses would encourage further adoption of this innovative approach toward farming practices - particularly given the increased demand for skilled professionals required during installation phases. Furthermore, given the potential benefits of BLCT technology for food security and agricultural sustainability, policymakers must develop supportive regulatory frameworks to promote investment in BLCT research and development.

## 7.3 Limitations and future research direction

Although the study provides insight into the intentions of using BLCT in AFSC, further research is required to address some limitations. Firstly, for the empirical study to find the ITU BLCT in AFSC, only the single SC decision maker perspective has been observed. Future research focusing on the dual perspective of the supplier- buyer's ITU the technology is recommended. Secondly, while transparency and traceability, food safety and security, and cost were deemed essential variables examined by this study, incorporating additional factors such as technical knowledge, government intervention, regulatory framework about governance, and policy implementation practices aimed at societal benefits are recommended avenues worthy of exploration. Thirdly, the model is tested on a sample of 258 working professionals in the Indian AFSC domain, thus restricting its generalizability across other countries unless supplemented by further studies.

Nevertheless, practitioners, researchers, and scholars can apply and extend our proven model to other nations through additional research. A comparative study between the developed and developing nations can be conducted to understand ITU BLCT. In addition, the food- and agriculture-

specific industries and country-specific ones may be examined to understand better the variables affecting or leading to adopting BLCT. Furthermore, a longitudinal study of practitioners' perceptions and the advanced industrial use cases would provide great insight into time-tested results.

# References

- Agrawal, P. and Narain, R. (2023), "Analysis of enablers for the digitalization of supply chain using an interpretive structural modelling approach", International Journal of Productivity and Performance Management, Vol. 72 No. 2, pp. 410-439. https://doi.org/10.1108/IJPPM-09-2020-0481
- Ahmed, W.A.H., MacCarthy, B.L. and Treiblmaier, H. (2022), "Why, where and how are organizations using blockchain in their supply chains? Motivations, application areas and contingency factors", *International Journal of Operations & Production Management*, Vol. 42 No. 12, pp. 1995–2028, doi: 10.1108/IJOPM-12-2021-0805.
- Ali, I. and Aboelmaged, M.G.S. (2022), "Implementation of supply chain 4.0 in the food and beverage industry: perceived drivers and barriers", International Journal of Productivity and Performance Management, Vol. 71 No. 4, pp. 1426–1443, doi: 10.1108/IJPPM-07-2020-0393
- Ali, I. and Govindan, K. (2023), "Extenuating operational risks through digital transformation of agri-food supply chains", *Production Planning & Control*, Vol. 34 No. 12, pp. 1165–1177, doi: 10.1080/09537287.2021.1988177.
- Ali, M.H., Chung, L., Kumar, A., Zailani, S. and Tan, K.H. (2021), "A sustainable Blockchain framework for the halal food supply chain: Lessons from Malaysia", Technological Forecasting and Social Change, Vol. 170, p. 120870, doi: 10.1016/j.techfore.2021.120870.
- Alladi, T., Chamola, V., Parizi, R. M., & Choo, K. K. R. (2019). Blockchain applications for industry 4.0 and industrial IoT: A review. *IEEE Access*, 7, pp. 176935-176951.
- Amoako-Gyampah, K. (2007), "Perceived usefulness, user involvement and behavioral intention: An empirical study of ERP implementation", *Computers in Human Behavior*, Vol. 23, No. 3, pp. 1232–1248. https://doi.org/10.1016/j.chb.2004.12.002
- Antonucci, F., Figorilli, S., Costa, C., Pallottino, F., Raso, L., & Menesatti, P. (2019), "A Review on blockchain applications in the agri-food sector", *Journal of the Science of Food and Agriculture*, Vol. 99, No. 14, pp. 6129-6138, doi: 10.1002/jsfa.9912.
- Armstrong, J. S., & Overton, T. S. (1977). Estimating Nonresponse Bias in Mail Surveys. *Journal of Marketing Research*, SAGE Publications, 14(3), 396–402.

- Asokan, D. R., Huq, F. A., Smith, C. M., & Stevenson, M. (2022), "Socially responsible operations in the Industry 4.0 era: post-COVID-19 technology adoption and perspectives on future research", *International Journal of Operations & Production Management*, Vol. 42 No. 13, pp. 185-217. https://doi.org/10.1108/IJOPM-01-2022-0069
- Astill, J., Dara, R.A., Campbell, M., Farber, J.M., Fraser, E.D.G., Sharif, S. and Yada, R.Y. (2019),
  "Transparency in food supply chains: A review of enabling technology solutions", *Trends in Food Science & Technology*, Vol. 91, pp. 240–247, doi: 10.1016/j.tifs.2019.07.024.
- Baharmand, H., Maghsoudi, A. and Coppi, G. (2021), "Exploring the application of blockchain to humanitarian supply chains: insights from Humanitarian Supply Blockchain pilot project", *International Journal of Operations & Production Management*, Vol. 41 No. 9, pp. 1522–1543, doi: 10.1108/IJOPM-12-2020-0884.
- Barbosa, M.W. (2021), "Uncovering research streams on agri-food supply chain management: A bibliometric study", *Global Food Security*, Vol. 28, p. 100517, doi: 10.1016/j.gfs.2021.100517.
- Cao, S., Powell, W., Foth, M., Natanelov, V., Miller, T. and Dulleck, U. (2021), "Strengthening consumer trust in beef supply chain traceability with a blockchain-based human-machine reconcile mechanism", Computers and Electronics in Agriculture, Vol. 180, p. 105886, doi: 10.1016/j.compag.2020.105886.
- Chang, S.E. and Chen, Y. (2020), "When Blockchain Meets Supply Chain: A Systematic Literature Review on Current Development and Potential Applications", IEEE Access, Vol. 8, pp. 62478–62494, doi: 10.1109/ACCESS.2020.2983601.
- Chang, Y., Iakovou, E. and Shi, W. (2020), "Blockchain in global supply chains and cross border trade: a critical synthesis of the state-of-the-art, challenges and opportunities", International Journal of Production Research, Vol. 58 No. 7, pp. 2082–2099, doi: 10.1080/00207543.2019.1651946.
- Chavalala, M.M., Bag, S., Pretorius, J.H.C. and Rahman, M.S. (2022), "A multi-method study on the barriers of the blockchain technology application in the cold supply chains", Journal of Enterprise Information Management, doi: 10.1108/JEIM-06-2022-0209.
- Chau, P. Y. (1997). Re-examining a model for evaluating information center success using a structural equation modelling approach. *Decision Sciences*, Vol. 28, No. 2, pp. 309-334.
- Chen, S., Liu, X., Yan, J., Hu, G. and Shi, Y. (2021), "Processes, benefits, and challenges for adoption of blockchain technologies in food supply chains: a thematic analysis", Information Systems

and E-Business Management, Vol. 19 No. 3, pp. 909–935, doi: 10.1007/s10257-020-00467-3.

- Chen, Y., Richter, J.I. and Patel, P.C. (2021), "Decentralized Governance of Digital Platforms", Journal of Management, Vol. 47 No. 5, pp. 1305–1337, doi: 10.1177/0149206320916755.
- Dash, G. and Paul, J. (2021), "CB-SEM vs PLS-SEM methods for research in social sciences and technology forecasting", Technological Forecasting and Social Change, Vol. 173, p. 121092, doi: 10.1016/j.techfore.2021.121092.
- Davis, F. D. (1985). A technology acceptance model for empirically testing new end-user information systems: Theory and results (Doctoral dissertation, Massachusetts Institute of Technology).
- Davis, F.D. (1989), "Perceived Usefulness, Perceived Ease of Use, and User Acceptance of Information Technology", MIS Quarterly, Vol. 13 No. 3, p. 319, doi: 10.2307/249008.
- Danese, P., Mocellin, R., & Romano, P. (2021). Designing blockchain systems to prevent counterfeiting in wine supply chains: a multiple-case study. *International Journal of Operations & Production Management*, Vol. 41, No. 13, pp. 1-33. https://doi.org/10.1108/IJOPM-12-2019-0781
- Duan, J., Zhang, C., Gong, Y., Brown, S. and Li, Z. (2020), "A Content-Analysis Based Literature Review in Blockchain Adoption within Food Supply Chain", International Journal of Environmental Research and Public Health, Vol. 17 No. 5, p. 1784, doi: 10.3390/ijerph17051784.
- Du, M., Chen, Q., Xiao, J., Yang, H., & Ma, X. (2020), "Supply chain finance innovation using blockchain", *IEEE Transactions on Engineering Management*, Vol. 67, No. 4, pp. 1045-1058. Doi: 10.1109/TEM.2020.2971858.
- Dwivedi, Y.K., Rana, N.P., Jeyaraj, A., Clement, M. and Williams, M.D. (2019), "Re-examining the Unified Theory of Acceptance and Use of Technology (UTAUT): Towards a Revised Theoretical Model", Information Systems Frontiers, Vol. 21 No. 3, pp. 719–734, doi: 10.1007/s10796-017-9774-y.
- Dwivedi, Y.K., Shareef, M.A., Simintiras, A.C., Lal, B. and Weerakkody, V. (2016), "A generalised adoption model for services: A cross-country comparison of mobile health (mhealth)", Government Information Quarterly, Vol. 33 No. 1, pp. 174–187, doi: 10.1016/j.giq.2015.06.003.

- FAO 2009, Global agriculture towards 2050 Food and Agriculture Organization https://www.fao.org/fileadmin/templates/wsfs/docs/Issues\_papers/HLEF2050\_Global\_Agric ulture.pdf (Assessed date 15-01-2023)
- Farshidi, S., Jansen, S., España, S., & Verkleij, J. (2020), "Decision support for blockchain platform selection: Three industry case studies", *IEEE Transactions on Engineering Management*, Vol. 67, No. 4, pp. 1109-1128. Doi: 10.1109/TEM.2019.2956897.
- Feng, H., Wang, X., Duan, Y., Zhang, J. and Zhang, X. (2020), "Applying blockchain technology to improve agri-food traceability: A review of development methods, benefits and challenges", *Journal of Cleaner Production*, Vol. 260, p. 121031, doi: 10.1016/j.jclepro.2020.121031.
- Fischer, C. (2013), "Trust and communication in European agri-food chains", Supply Chain Management: An International Journal, Vol.18, No. 2, pp. 208-218. https://doi.org/10.1108/13598541311318836
- Fornell, C. and Larcker, D.F. (1981), "Structural Equation Models with Unobservable Variables and Measurement Error: Algebra and Statistics", *Journal of Marketing Research*, Vol. 18 No. 3, p. 382, doi: 10.2307/3150980.
- Ganbold, O., Matsui, Y. and Rotaru, K. (2021), "Effect of information technology-enabled supply chain integration on firm's operational performance", *Journal of Enterprise Information Management*, Vol. 34 No. 3, pp. 948–989, doi: 10.1108/JEIM-10-2019-0332.
- Galvez, J.F., Mejuto, J.C. and Simal-Gandara, J. (2018), "Future challenges on the use of blockchain for food traceability analysis", *TrAC Trends in Analytical Chemistry*, Vol. 107, pp. 222–232, doi: 10.1016/j.trac.2018.08.011.
- Gardas, B.B., Raut, R.D. and Narkhede, B. (2017), "Modeling causal factors of post-harvesting losses in vegetable and fruit supply chain: An Indian perspective", Renewable and Sustainable Energy Reviews, Vol. 80, pp. 1355–1371, doi: 10.1016/j.rser.2017.05.259.
- Gardas, B., Raut, R., Jagtap, A. H., & Narkhede, B. (2019), "Exploring the key performance indicators of green supply chain management in agro-industry", *Journal of Modelling in Management, Vol.* 14, No. 1, pp. 260–283. https://doi.org/10.1108/JM2-12-2017-0139
- Guggenberger, T., Schweizer, A., & Urbach, N. (2020). Improving interorganizational information sharing for vendor managed inventory: Toward a decentralized information hub using blockchain technology. *IEEE Transactions on Engineering Management*, Vol 67, No. 4, pp. 1074-1085. Doi: 10.1109/TEM.2020.2978628.

- Gupta, R., & Shankar, R. (2023), "Managing food security using blockchain-enabled traceability system", *Benchmarking: An International Journal*. https://doi.org/10.1108/BIJ-01-2022-0029
- Hassani, H., Huang, X. and Silva, E. (2018), "Banking with blockchain-ed big data", Journal of Management Analytics, Vol. 5 No. 4, pp. 256–275, doi: 10.1080/23270012.2018.1528900.
- Hew, J.-J., Wong, L.-W., Tan, G.W.-H., Ooi, K.-B. and Lin, B. (2020), "The blockchain-based Halal traceability systems: a hype or reality?", Supply Chain Management: An International Journal, Vol. 25 No. 6, pp. 863–879, doi: 10.1108/SCM-01-2020-0044.
- Iftekhar, A., Cui, X., Hassan, M., & Afzal, W. (2020), "Application of blockchain and Internet of Things to ensure tamper-proof data availability for food safety", *Journal of Food Quality*, Vol. 2020. https://doi.org/10.1155/2020/5385207
- (IBM, 2017), SPSS Amos | IBM". (n.d.)., available at: https://www.ibm.com/products/structuralequation-modeling-sem (accessed 12 February 2024).
- Jiang, X., Wang, G., De Clercq, D., & Yi, X. (2020), "How Do Firms Achieve Successful Technology Commercialization? Evidence from Chinese Manufacturing Firms", *IEEE Transactions on Engineering Management*. Vol. 69, No. 5, pp. 1973-1986. Doi: 10.1109/TEM.2020.2997965.
- Jiang, S., Li, Y., Wang, S. and Zhao, L. (2022), "Blockchain competition: The tradeoff between platform stability and efficiency", European Journal of Operational Research, Vol. 296 No. 3, pp. 1084–1097, doi: 10.1016/j.ejor.2021.05.031.
- Kafetzopoulos, D., Margariti, S., Stylios, C., Arvaniti, E., & Kafetzopoulos, P. (2023), "Managing the traceability system for food supply chain performance", *International Journal of Productivity and Performance Management*. https://doi.org/10.1108/IJPPM-12-2021-0690
- Kamath, R. (2018), "Food Traceability on Blockchain: Walmart's Pork and Mango Pilots with IBM", The Journal of the British Blockchain Association, Vol. 1 No. 1, pp. 1–12, doi: 10.31585/jbba-1-1-(10)2018.
- Kamble, S., Gunasekaran, A. and Arha, H. (2019), "Understanding the Blockchain technology adoption in supply chains-Indian context", International Journal of Production Research, Vol. 57 No. 7, pp. 2009–2033, doi: 10.1080/00207543.2018.1518610.
- Kamilaris, A., Fonts, A. and Prenafeta-Boldú, F.X. (2019), "The rise of blockchain technology in agriculture and food supply chains", Trends in Food Science & Technology, Vol. 91, pp. 640– 652, doi: 10.1016/j.tifs.2019.07.034.
- Karamchandani, A., Srivastava, S.K., Kumar, S. and Srivastava, A. (2021), "Analysing perceived role of blockchain technology in SCM context for the manufacturing industry", International

Journal of Production Research, Vol. 59 No. 11, pp. 3398–3429, doi: 10.1080/00207543.2021.1883761.

- Karakas, S., Acar, A.Z. and Kucukaltan, B. (2021), "Blockchain adoption in logistics and supply chain: a literature review and research agenda", International Journal of Production Research, pp. 1–24, doi: 10.1080/00207543.2021.2012613.
- Kayikci, Y., Subramanian, N., Dora, M. and Bhatia, M.S. (2022), "Food supply chain in the era of Industry 4.0: blockchain technology implementation opportunities and impediments from the perspective of people, process, performance, and technology", Production Planning & Control, Vol. 33 No. 2–3, pp. 301–321, doi: 10.1080/09537287.2020.1810757.
- Khan, S., Kaushik, M.K., Kumar, R. and Khan, W. (2023), "Investigating the barriers of blockchain technology integrated food supply chain: a BWM approach", Benchmarking: An International Journal, Vol. 30 No. 3, pp. 713–735, doi: 10.1108/BIJ-08-2021-0489.
- Ko, T., Lee, J. and Ryu, D. (2018), "Blockchain Technology and Manufacturing Industry: Real-Time Transparency and Cost Savings", Sustainability, Vol. 10 No. 11, p. 4274, doi: 10.3390/su10114274.
- Krzyzanowski Guerra, K. and Boys, K.A. (2022), "A new food chain: Adoption and policy implications to blockchain use in agri-food industries", Applied Economic Perspectives and Policy, Vol. 44 No. 1, pp. 324–349, doi: 10.1002/aepp.13163.
- Kshetri, N. (2019), "Blockchain and the Economics of Food Safety", IT Professional, Vol. 21 No. 3, pp. 63–66, doi: 10.1109/MITP.2019.2906761.
- Kumar, S., Raut, R.D., Priyadarshinee, P., Mangla, S.K., Awan, U. and Narkhede, B.E. (2022), "The Impact of IoT on the Performance of Vaccine Supply Chain Distribution in the COVID-19 Context", IEEE Transactions on Engineering Management, pp. 1–11, doi: 10.1109/TEM.2022.3157625.
- Kumar, M., Choubey, V.K., Raut, R.D. and Jagtap, S. (2023a), "Enablers to achieve zero hunger through IoT and blockchain technology and transform the green food supply chain systems", Journal of Cleaner Production, Vol. 405, p. 136894, doi: 10.1016/j.jclepro.2023.136894.
- Kumar, M., Raut, R.D., Mangla, S.K., Chowdhury, S. and Choubey, V.K. (2024), "Moderating ESG compliance between industry 4.0 and green practices with green servitization: Examining its impact on green supply chain performance", Technovation, Vol. 129, p. 102898, doi: 10.1016/j.technovation.2023.102898.

- Kumar, V., Yetkin Ekren, B., Wang, J., Shah, B., & Frederico, G. F. (2023), "Investigating the impact of COVID-19 on sustainable food supply chains", *Journal of Modelling in Management*, Vol.18, No.4, pp. 1250–1273. https://doi.org/10.1108/JM2-03-2022-0072
- Lakhani, K. R., & Iansiti, M. (2017), "The truth about blockchain", *Harvard Business Review*, Vol. 95, No.1, pp. 119-127.
- Lanzolla, G. and Suarez, F.F. (2012), "Closing the Technology Adoption–Use Divide: The Role of Contiguous User Bandwagon", Journal of Management, Vol. 38 No. 3, pp. 836–859, doi: 10.1177/0149206310369938.
- Li, G., Xue, J., Li, N. and Ivanov, D. (2022), "Blockchain-supported business model design, supply chain resilience, and firm performance", Transportation Research Part E: Logistics and Transportation Review, Vol. 163, p. 102773, doi: 10.1016/j.tre.2022.102773.
- Li, K., Lee, J.-Y. and Gharehgozli, A. (2023), "Blockchain in food supply chains: a literature review and synthesis analysis of platforms, benefits and challenges", International Journal of Production Research, Vol. 61 No. 11, pp. 3527–3546, doi: 10.1080/00207543.2021.1970849.
- Liu, W., Shao, X.-F., Wu, C.-H. and Qiao, P. (2021), "A systematic literature review on applications of information and communication technologies and blockchain technologies for precision agriculture development", Journal of Cleaner Production, Vol. 298, p. 126763, doi: 10.1016/j.jclepro.2021.126763.
- Liu, W., Wang, J., Jia, F. and Choi, T.-M. (2022), "Blockchain announcements and stock value: a technology management perspective", International Journal of Operations & Production Management, Vol. 42 No. 5, pp. 713–742, doi: 10.1108/IJOPM-08-2021-0534.
- Luarn, P. and Lin, H.-H. (2005), "Toward an understanding of the behavioral intention to use mobile banking", Computers in Human Behavior, Vol. 21 No. 6, pp. 873–891, doi: 10.1016/j.chb.2004.03.003.
- Mangla, S.K., Kazancoglu, Y., Ekinci, E., Liu, M., Özbiltekin, M. and Sezer, M.D. (2021), "Using system dynamics to analyze the societal impacts of blockchain technology in milk supply chainsrefer", Transportation Research Part E: Logistics and Transportation Review, Vol. 149, p. 102289, doi: 10.1016/j.tre.2021.102289.
- Marucheck, A., Greis, N., Mena, C. and Cai, L. (2011), "Product safety and security in the global supply chain: Issues, challenges and research opportunities", Journal of Operations Management, Vol. 29 No. 7–8, pp. 707–720, doi: 10.1016/j.jom.2011.06.007.
- Masudin, I., Ramadhani, A., Restuputri, D.P. and Amallynda, I. (2021), "The Effect of Traceability System and Managerial Initiative on Indonesian Food Cold Chain Performance: A Covid-19

Pandemic Perspective", Global Journal of Flexible Systems Management, Vol. 22 No. 4, pp. 331–356, doi: 10.1007/s40171-021-00281-x.

- Mollenkopf, D.A., Peinkofer, S.T. and Chu, Y.( J. (2022), "Supply chain transparency: Consumer reactions to incongruent signals", Journal of Operations Management, Vol. 68 No. 4, pp. 306– 327, doi: 10.1002/joom.1180.
- Mondal, S., Wijewardena, K.P., Karuppuswami, S., Kriti, N., Kumar, D. and Chahal, P. (2019),
  "Blockchain Inspired RFID-Based Information Architecture for Food Supply Chain", IEEE
  Internet of Things Journal, Vol. 6 No. 3, pp. 5803–5813, doi: 10.1109/JIOT.2019.2907658.
- Mukherjee, A.A., Singh, R.K., Mishra, R. and Bag, S. (2022), "Application of blockchain technology for sustainability development in agricultural supply chain: justification framework", Operations Management Research, Vol. 15 No. 1–2, pp. 46–61, doi: 10.1007/s12063-021-00180-5.
- Mukri, B. (2018), "Blockchain technology in supply chain management: a review", *International Research Journal of Engineering and Technology*, Vol. 5, No. (6), pp. 2497-2500.
- Nayal, K., Raut, R.D., Yadav, V.S., Priyadarshinee, P. and Narkhede, B.E. (2022), "The impact of sustainable development strategy on sustainable supply chain firm performance in the digital transformation era", Business Strategy and the Environment, Vol. 31 No. 3, pp. 845–859, doi: 10.1002/bse.2921.
- Nayal, K., Raut, R.D., Narkhede, B.E., Priyadarshinee, P., Panchal, G.B. and Gedam, V.V. (2023), "Antecedents for blockchain technology-enabled sustainable agriculture supply chain", Annals of Operations Research, Vol. 327 No. 1, pp. 293–337, doi: 10.1007/s10479-021-04423-3.
- Paul, T., Mondal, S., Islam, N. and Rakshit, S. (2021), "The impact of blockchain technology on the tea supply chain and its sustainable performance", Technological Forecasting and Social Change, Vol. 173, p. 121163, doi: 10.1016/j.techfore.2021.121163.
- Podsakoff, P.M., MacKenzie, S.B., Lee, J.-Y. and Podsakoff, N.P. (2003), "Common method biases in behavioral research: A critical review of the literature and recommended remedies.", Journal of Applied Psychology, Vol. 88 No. 5, pp. 879–903, doi: 10.1037/0021-9010.88.5.879.
- Podsakoff, P.M. and Organ, D.W. (1986), "Self-Reports in Organizational Research: Problems and Prospects", Journal of Management, Vol. 12 No. 4, pp. 531–544, doi: 10.1177/014920638601200408.

- Queiroz, M.M. and Fosso Wamba, S. (2019), "Blockchain adoption challenges in supply chain: An empirical investigation of the main drivers in India and the USA", International Journal of Information Management, Vol. 46, pp. 70–82, doi: 10.1016/j.ijinfomgt.2018.11.021.
- Queiroz, M.M., Fosso Wamba, S., De Bourmont, M. and Telles, R. (2021), "Blockchain adoption in operations and supply chain management: empirical evidence from an emerging economy", International Journal of Production Research, Vol. 59 No. 20, pp. 6087–6103, doi: 10.1080/00207543.2020.1803511.
- Rana, N.P., Dwivedi, Y.K., Lal, B., Williams, M.D. and Clement, M. (2017), "Citizens' adoption of an electronic government system: towards a unified view", Information Systems Frontiers, Vol. 19 No. 3, pp. 549–568, doi: 10.1007/s10796-015-9613-y.
- Rana, N.P., Dwivedi, Y.K., Williams, M.D. and Weerakkody, V. (2016), "Adoption of online public grievance redressal system in India: Toward developing a unified view", Computers in Human Behavior, Vol. 59, pp. 265–282, doi: 10.1016/j.chb.2016.02.019.
- Roeck, D., Sternberg, H. and Hofmann, E. (2020), "Distributed ledger technology in supply chains: a transaction cost perspective", International Journal of Production Research, Vol. 58 No. 7, pp. 2124–2141, doi: 10.1080/00207543.2019.1657247.
- Rogerson, M. and Parry, G.C. (2020), "Blockchain: case studies in food supply chain visibility", Supply Chain Management: An International Journal, Vol. 25 No. 5, pp. 601–614, doi: 10.1108/SCM-08-2019-0300.
- Ronaghi, M.H. (2021), "A blockchain maturity model in agricultural supply chain", Information Processing in Agriculture, Vol. 8 No. 3, pp. 398–408, doi: 10.1016/j.inpa.2020.10.004.
- Roy, V. and Srivastava, S.K. (2022), "The safety-quality dominant view of food chain integrity: Implications for consumer-centric food chain governance", International Journal of Management Reviews, Vol. 24 No. 1, pp. 3–24, doi: 10.1111/ijmr.12258.
- Saha, A., Raut, R. and Kumar, M. (2023), Digital technology adoption challenges in the agri-food supply chain from the perspective of attaining sustainable development goals, *The International Journal of Logistics Management*. https://doi.org/10.1108/IJLM-09-2023-0412
- Saha, A., Raut, R.D. and Kumar, M. (2023a), Leveraging Blockchain Technology to Combat Food Fraud in the Agri-food Supply Chain. Int J Food Sci Technol. Accepted Author Manuscript. https://doi.org/10.1111/ijfs.16889
- Salah, K., Nizamuddin, N., Jayaraman, R. and Omar, M. (2019), "Blockchain-Based Soybean Traceability in Agricultural Supply Chain", IEEE Access, Vol. 7, pp. 73295–73305, doi: 10.1109/ACCESS.2019.2918000.

- Saurabh, K., Rani, N. and Upadhyay, P. (2023), "Towards blockchain led decentralized autonomous organization (DAO) business model innovations", Benchmarking: An International Journal, Vol. 30 No. 2, pp. 475–502, doi: 10.1108/BIJ-10-2021-0606.
- Saurabh, S. and Dey, K. (2021), "Blockchain technology adoption, architecture, and sustainable agrifood supply chains", Journal of Cleaner Production, Vol. 284, p. 124731, doi: 10.1016/j.jclepro.2020.124731.
- Schmidt, C.G. and Wagner, S.M. (2019), "Blockchain and supply chain relations: A transaction cost theory perspective", Journal of Purchasing and Supply Management, Vol. 25 No. 4, p. 100552, doi: 10.1016/j.pursup.2019.100552.
- Schulze-Ehlers, B., Steffen, N., Busch, G. and Spiller, A. (2014), "Supply chain orientation in SMEs as an attitudinal construct: Conceptual considerations and empirical application to the dairy sector", Supply Chain Management: An International Journal, Vol. 19 No. 4, pp. 395–412, doi: 10.1108/SCM-07-2013-0241.
- Shah, R. and Goldstein, S.M. (2006), "Use of structural equation modeling in operations management research: Looking back and forward", Journal of Operations Management, Vol. 24 No. 2, pp. 148–169, doi: 10.1016/j.jom.2005.05.001.
- Sharma, M. G. (2023), "Supply chain, geographical indicator and blockchain: Provenance model for commodity", *International Journal of Productivity and Performance Management*, Vol 72, No. 1, pp. 92–108. https://doi.org/10.1108/IJPPM-05-2021-0288
- Shi, J. and Jiang, Z. (2023), "Willingness to pay a premium price for green products: does a reference group matter?", Environment, Development and Sustainability, Vol. 25 No. 8, pp. 8699– 8727, doi: 10.1007/s10668-022-02419-y.
- Srivastava, A. and Dashora, K. (2022), "Application of blockchain technology for agrifood supply chain management: a systematic literature review on benefits and challenges", Benchmarking: An International Journal, Vol. 29 No. 10, pp. 3426–3442, doi: 10.1108/BIJ-08-2021-0495.
- Sternberg, H.S., Hofmann, E. and Roeck, D. (2021), "The Struggle is Real: Insights from a Supply Chain Blockchain Case", Journal of Business Logistics, Vol. 42 No. 1, pp. 71–87, doi: 10.1111/jbl.12240.
- Susanty, A., Puspitasari, N. B., Prastawa, H., & Renaldi, S. V. (2021), "Exploring the best policy scenario plan for the dairy supply chain: A DEMATEL approach", *Journal of Modelling in Management*, Vol. 16, No. 1, pp. 240–266. https://doi.org/10.1108/JM2-08-2019-0185
- Sunny, J., Undralla, N. and Madhusudanan Pillai, V. (2020), "Supply chain transparency through blockchain-based traceability: An overview with demonstration", Computers & Industrial Engineering, Vol. 150, p. 106895, doi: 10.1016/j.cie.2020.106895.
- Tabachnick, B. G., Fidell, L. S., & Ullman, J. B. (2007), "Using Multivariate Statistics" (Vol. 5, pp. 481-498). Boston, MA: Pearson.
- Teo, A. C., Tan, G. W. H., Ooi, K. B., Hew, T. S., & Yew, K. T. (2015), "The effects of convenience and speed in m-payment", *Industrial Management & Data Systems*, Vol. 115 No. 2, pp. 311-331).
- Tian, F. (2016) An Agri-Food Supply Chain Traceability System for China Based on RFID & Blockchain Technology 13th International Conference on Service Systems and Service Management. Kunming, China, 24-26 June 2016 Beijing: China.
- Tran, L.T.T. and Nguyen, P.T. (2021), "Co-creating blockchain adoption: theory, practice and impact on usage behavior", Asia Pacific Journal of Marketing and Logistics, Vol. 33 No. 7, pp. 1667– 1684, doi: 10.1108/APJML-08-2020-0609.
- Treiblmaier, H. (2018), "The Impact of the Blockchain on the Supply Chain: A Theory-Based Research Framework and a Call for Action", SSRN Electronic Journal, doi: 10.2139/ssrn.3224145.
- Trollman, H., Garcia-Garcia, G., Jagtap, S., & Trollman, F. (2022), "Blockchain for ecologically embedded coffee supply chains", Logistics, Vol. 6, No. 3, pp. 43. https://doi.org/10.3390/logistics6030043
- Ullman, J. B., & Bentler, P. M. (2012). Structural equation modelling. In Handbook of Psychology (2nd ed. Vol. 2).
- Vasan, M. and Yoganandan, G. (2023), "Does the belief of farmers on land as God influence the adoption of smart farming technologies?", Benchmarking: An International Journal, doi: 10.1108/BIJ-10-2022-0645.
- Venkatesh, V. (2022), "Adoption and use of AI tools: a research agenda grounded in UTAUT", Annals of Operations Research, Vol. 308 No. 1–2, pp. 641–652, doi: 10.1007/s10479-020-03918-9.
- Venkatesh, V. and Bala, H. (2008), "Technology Acceptance Model 3 and a Research Agenda on Interventions", Decision Sciences, Vol. 39 No. 2, pp. 273–315, doi: 10.1111/j.1540-5915.2008.00192.x.

- Venkatesh, V. and Davis, F.D. (2000), "A Theoretical Extension of the Technology Acceptance Model: Four Longitudinal Field Studies", Management Science, Vol. 46 No. 2, pp. 186–204, doi: 10.1287/mnsc.46.2.186.11926.
- Venkatesh, Morris, Davis, and Davis. (2003), "User Acceptance of Information Technology: Toward a Unified View", MIS Quarterly, Vol. 27 No. 3, p. 425, doi: 10.2307/30036540.
- Venkatesh, Thong, and Xu. (2012), "Consumer Acceptance and Use of Information Technology: Extending the Unified Theory of Acceptance and Use of Technology", MIS Quarterly, Vol. 36 No. 1, p. 157, doi: 10.2307/41410412.
- Vikaliana, R., Rasi, R.Z.R.M. and Pujawan, I.N. (2021), "Traceability System on Mangosteen Supply Chain Management Using Blockchain Technology: A Model Design", Studies of Applied Economics, Vol. 39 No. 4, doi: 10.25115/eea.v39i4.4565.
- Villena, V.H., Choi, T.Y. and Revilla, E. (2019), "Revisiting Interorganizational Trust: Is More Always Better or Could More Be Worse?", Journal of Management, Vol. 45 No. 2, pp. 752– 785, doi: 10.1177/0149206316680031.
- Vu, N., Ghadge, A. and Bourlakis, M. (2023), "Blockchain adoption in food supply chains: a review and implementation framework", Production Planning & Control, Vol. 34 No. 6, pp. 506– 523, doi: 10.1080/09537287.2021.1939902.
- Wamba, S.F. and Queiroz, M.M. (2019), "The Role of Social Influence in Blockchain Adoption: The Brazilian Supply Chain Case", IFAC-PapersOnLine, Vol. 52 No. 13, pp. 1715–1720, doi: 10.1016/j.ifacol.2019.11.448.
- Wamba, S.F. and Queiroz, M.M. (2022), "Industry 4.0 and the supply chain digitalisation: a blockchain diffusion perspective", Production Planning & Control, Vol. 33 No. 2–3, pp. 193– 210, doi: 10.1080/09537287.2020.1810756.
- Wamba, S.F., Queiroz, M.M. and Trinchera, L. (2020), "Dynamics between blockchain adoption determinants and supply chain performance: An empirical investigation", International Journal of Production Economics, Vol. 229, p. 107791, doi: 10.1016/j.ijpe.2020.107791.
- Wang, H., Zhang, M., Ying, H. and Zhao, X. (2021), "The impact of blockchain technology on consumer behavior: a multimethod study", Journal of Management Analytics, Vol. 8 No. 3, pp. 371–390, doi: 10.1080/23270012.2021.1958264.
- Wang, Y., Han, J.H. and Beynon-Davies, P. (2019a), "Understanding blockchain technology for future supply chains: a systematic literature review and research agenda", Supply Chain Management: An International Journal, Vol. 24 No. 1, pp. 62–84, doi: 10.1108/SCM-03-2018-0148.

- Wang, Y., Singgih, M., Wang, J. and Rit, M. (2019b), "Making sense of blockchain technology: How will it transform supply chains?", International Journal of Production Economics, Vol. 211, pp. 221–236, doi: 10.1016/j.ijpe.2019.02.002.
- Wang, O. and Scrimgeour, F. (2023), "Consumer adoption of blockchain food traceability: effects of innovation-adoption characteristics, expertise in food traceability and blockchain technology, and segmentation", British Food Journal, Vol. 125 No. 7, pp. 2493–2513, doi: 10.1108/BFJ-06-2022-0466.
- Warshaw, P. R., & Davis, F. D. (1985), "Disentangling intention and expectation", Journal of Experimental Social Psychology, Vol. 21, No. 3, pp. 213-228.
- Wong, L.-W., Leong, L.-Y., Hew, J.-J., Tan, G.W.-H. and Ooi, K.-B. (2020a), "Time to seize the digital evolution: Adoption of blockchain in operations and supply chain management among Malaysian SMEs", International Journal of Information Management, Vol. 52, p. 101997, doi: 10.1016/j.ijinfomgt.2019.08.005.
- Wong, L.-W., Tan, G.W.-H., Lee, V.-H., Ooi, K.-B. and Sohal, A. (2020b), "Unearthing the determinants of Blockchain adoption in supply chain management", International Journal of Production Research, Vol. 58 No. 7, pp. 2100–2123, doi: 10.1080/00207543.2020.1730463.
- Xu, J., Guo, S., Xie, D. and Yan, Y. (2020), "Blockchain: A new safeguard for agri-foods", Artificial Intelligence in Agriculture, Vol. 4, pp. 153–161, doi: 10.1016/j.aiia.2020.08.002.
- Yadav, V.S., Singh, A.R., Raut, R.D. and Govindarajan, U.H. (2020), "Blockchain technology adoption barriers in the Indian agricultural supply chain: an integrated approach", Resources, Conservation and Recycling, Vol. 161, p. 104877, doi: 10.1016/j.resconrec.2020.104877.
- Yen, D. C., Wu, C.-S., Cheng, F.-F., & Huang, Y.-W. (2010), "Determinants of users' intention to adopt wireless technology: An empirical study by integrating TTF with TAM", *Computers in Human Behavior*, Vol. 26, No. 5, pp. 906–915. https://doi.org/10.1016/j.chb.2010.02.005
- Yiannas, F. (2018), "A New Era of Food Transparency Powered by Blockchain", Innovations: Technology, Governance, Globalization, Vol. 12 No. 1–2, pp. 46–56, doi: 10.1162/inov a 00266.
- Zeng, M. and Lu, J. (2021), "The impact of information technology capabilities on agri-food supply chain performance: the mediating effects of interorganizational relationships", Journal of Enterprise Information Management, Vol. 34 No. 6, pp. 1699–1721, doi: 10.1108/JEIM-08-2019-0237.

- Zhang, X., Sun, P., Xu, J., Wang, X., Yu, J., Zhao, Z. and Dong, Y. (2020), "Blockchain-Based Safety Management System for the Grain Supply Chain", IEEE Access, Vol. 8, pp. 36398–36410, doi: 10.1109/ACCESS.2020.2975415.
- Zhao, Guoqing; Liu, Shaofeng; Lopez, Carmen; Lu, Haiyan; Elgueta, Sebastian; Chen, Huilan; Boshkoska, Biljana Mileva (2019). Blockchain technology in agri-food value chain management: A synthesis of applications, challenges and future research directions. Computers in Industry, Vol. 109, pp. 83–99.

## Annexure



A1 – Path analysis diagram of the conceptual framework